UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

DATE OF REPORT (DATE OF EARLIEST EVENT REPORTED): JANUARY 7, 2004

OSI SYSTEMS, INC.

(EXACT NAME OF REGISTRANT SPECIFIED IN CHARTER)

N)

000-23125 (COMMISSION FILE NUMBER) 330238801 (IRS EMPLOYER IDENTIFICATION NO.)

12525 CHADRON AVENUE HAWTHORNE, CA 90250

(ADDRESS OF PRINCIPAL EXECUTIVE OFFICES) (ZIP CODE)

(310) 978-0516 (REGISTRANT'S TELEPHONE NUMBER, INCLUDING AREA CODE)

CALIFORNIA (STATE OF INCORPORATION)

Item 2. Acquisition or Disposition of Assets

On January 7, 2004, OSI Systems, Inc. ("OSI Systems"), completed the acquisition of Advance Research & Applications Corp., a California corporation ("ARACOR") pursuant to a certain Merger Agreement and Plan of Organization dated as of December 18, 2003 (the "Merger Agreement"). The acquisition was accomplished by means of a merger between OSI Subsidiary, Inc., a California corporation and a wholly owned subsidiary of OSI Systems ("OSI Subsidiary"), with and into ARACOR.

Under the terms of the Merger Agreement, each issued and outstanding share of common stock and each option to purchase common stock (whether vested or not) of ARACOR was converted into the right to receive from OSI Systems: (i) an initial cash payment ("Initial Cash Payment") calculated in accordance with the following formula: (\$19,601,000.00 plus the aggregate of all exercise prices payable with respect to ARACOR's issued and outstanding options to purchase common stock of ARACOR as of the closing date of the merger and minus transaction legal and financial advisory fees) divided by (the number of issued and outstanding shares of ARACOR's common stock as of the closing date of the merger plus the number of shares of ARACOR's common stock issuable upon exercise of ARACOR's issued and outstanding options as of the closing date of the merger); and (ii) a contingent cash payment (the "Contingent Cash Payment") calculated based on ARACOR's revenues, net sales and other financial measures more fully described in Section 4 of the Merger Agreement.

After application of the Initial Cash Payment formula described above, payments to ARACOR's shareholders and optionholders totaled \$18,972,394.00 in the aggregate. The total amount of the Contingent Cash Payment payable to ARACOR's shareholders and optionholders shall not exceed \$30,000,000.00 in the aggregate. Each Contingent Cash Payment is due 90 days after June 30 of each year included in the contingent payment period. The last Contingent Cash Payment is due 90 days after December 31, 2010, if applicable. The aggregate amount of the Initial Cash Payment was paid to the ARACOR's shareholders on a pro rata basis and to ARACOR's optionholders, also on a pro rata basis, but after adjustment to account for the exercise price of their options. The Contingent Cash Payment shall be paid to both ARACOR's shareholders and optionholders on a pro rata basis. OSI Systems used its available cash to make the Initial Cash Payment and intends to use its cash on hand to make the Contingent Cash Payments.

Following the close of the Merger Agreement, all of ARACOR's plant and equipment will continue to be operated in their current location in Sunnyvale, California.

ARACOR develops and manufactures X-ray test and inspection systems for industrial nondestructive evaluation and process control applications; manufactures cargo inspection systems; and provides research services in related areas. ARACOR is a leading manufacturer of digital radiographic ("DR") and computed tomography ("CT") systems.

ARACOR is a leading U.S. manufacturer of high-energy X-ray imaging systems. The X-ray imaging products include DR, real-time radiography, and CT systems, which are used for industrial nondestructive inspection, quality control, metrology and for the non-intrusive inspection of vehicles and cargo containers to detect contraband. ARACOR also manufactures the Model 4100 Automatic Semiconductor Irradiation System used for process control and testing of electronic components.

ARACOR's patented Model 4100 Automatic Semiconductor Irradiation System is used to test military and aerospace electronic components at the wafer level of production, to determine if they meet specifications for survivability to total radiation exposure and transient-radiation/latch-up effects. The Model 4100 is the only instrument available that can perform these types of tests at the wafer level and in a general laboratory environment. Significant cost and throughput advantages are realized with the use of this instrument.

Prior to the Merger Agreement, ARACOR purchased photodiodes from UDT Sensors, Inc., a wholly owned subsidiary of OSI Systems.

Item 7. Financial Statements and Exhibits

a. The financial statements of the business acquired will be filed by amendment within the allowable time period.

b. The pro forma financial information will be included in the amendment referred to above.

c. Exhibits:

2.1 Merger Agreement and Plan of Organization dated as of December 18, 2003 by and among ARACOR, Robert A. Armistead, OSI Subsidiary, Inc., and OSI.

Exhibit Index

2.1 Merger Agreement and Plan of Organization dated as of December 18, 2003 by and among ARACOR, Robert A. Armistead, OSI Subsidiary, Inc., and OSI.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

OSI SYSTEMS, INC.

Date: January 21, 2004

By: /s/ Victor Sze

Victor Sze Secretary and General Counsel

MERGER AGREEMENT AND PLAN OF REORGANIZATION

THIS MERGER AGREEMENT AND PLAN OF REORGANIZATION (this "<u>Agreement</u>") is dated as of December 18, 2003, and is entered into by and among Advanced Research & Applications Corporation, a California corporation ("<u>ARACOR</u>"), Robert A. Armistead ("<u>Individual</u>"), OSI Subsidiary, Inc., a California corporation ("<u>OSI Subsidiary</u>"), OSI Systems, Inc., a California corporation ("<u>OSI Subsidiary</u>"), O

A. OSI and OSI Subsidiary wish to acquire the Business (as hereinafter defined) by merging OSI Subsidiary into ARACOR, of which Individual is the largest shareholder.

B. The parties are entering into this Agreement in order to accomplish such a merger.

Accordingly, the parties to this Agreement hereby agree as follows:

1. CERTAIN DEFINITIONS. As used in this agreement, the following capitalized terms shall have the meanings set forth in this Section 1.

1.1. <u>Affiliate</u>. "Affiliate" means, with respect to a particular Person, any other Person as to which the particular Person directly, or indirectly through one or more intermediaries, controls, is controlled by or is under common control with. For purposes of this definition, "control" means, with respect to a Person that is a corporation, the right to exercise, directly or indirectly, more than fifty percent (50%) of the voting rights attributable to the shares of the controlled corporation and, with respect to a Person that is not a corporation, the possession, directly or indirectly or indirectly of the management or policies of the controlled Person.

1.2. <u>ARACOR Common Stock</u>. "ARACOR Common Stock" means all the issued and outstanding Common Stock of ARACOR, \$0.001 par value per share.

1.3. <u>ARACOR Disclosure Schedule</u>. "ARACOR Disclosure Schedule" means the disclosure schedule delivered by ARACOR to OSI concurrently with ARACOR's execution of this Agreement, together with any updates provided as of the Closing Date.

1.4. ARACOR Documents. "ARACOR Documents" shall have the meaning ascribed to it in Section 6.2 hereof.

1.5. ARACOR Employee. "ARACOR Employee" shall have the meaning ascribed to it in Section 9.6 hereof.

1.6. ARACOR Financial Statements. "ARACOR Financial Statements" shall have the meaning ascribed to it in Section 6.5.1 hereof.

1.7. ARACOR Intellectual Property. "ARACOR Intellectual Property" shall have the meaning ascribed to it in Section 6.17.1 hereof.

1.8. <u>ARACOR's Knowledge</u>. "ARACOR's Knowledge" means knowledge that either (a) was in the actual possession of ARACOR's executive officers, or any of them; or (b) should have been in the possession of ARACOR's executive officers, or any or them, based on their status and duties as executive officers of ARACOR.

1

1.9. **ARACOR Optionholder**. "ARACOR Optionholder" means, as of the time immediately preceding the Effective Time, all holders of options to purchase ARACOR Common Stock.

1.10. **ARACOR Optionholder Exercise Adjustment**. "ARACOR Optionholder Exercise Adjustment" shall mean the aggregate amount of consideration received or receivable by ARACOR upon the exercise of ARACOR Stock Options for the period between the date of this Agreement and the Effective Time.

1.11. ARACOR Service. "ARACOR Service" shall have the meaning ascribed to it in Section 9.6 hereof.

1.12. <u>ARACOR Shareholders</u>. "ARACOR Shareholders" means, as of the Effective Time, all holders of ARACOR Common Stock (inclusive of Individual), and "ARACOR Shareholder" means a single holder of ARACOR Common Stock.

1.13. ARACOR Software. "ARACOR Software" shall have the meaning ascribed to it in Section 6.17.1 hereof.

1.14. **ARACOR Stock Options**. "ARACOR Stock Options" means all outstanding options (whether or not vested) to purchase shares of ARACOR Common Stock heretofore granted under any stock option or other stock-based incentive plan, program or arrangement of ARACOR.

1.15. ARACOR Trademarks. "ARACOR Trademarks" shall have the meaning ascribed to it in Section 6.17.1 hereof.

1.16. ARACOR Trade Secrets. "ARACOR Trade Secrets" shall have the meaning ascribed to it in Section 6.17.1 hereof.

1.17. <u>Assets</u>. "Assets" means all of the equipment, inventory and other assets and properties of ARACOR or that ARACOR owns or leases or utilizes in the Business.

1.18. Audit Period Termination Date. "Audit Period Termination Date" shall have the meaning ascribed to it in Section 4.6 hereof.

1.19. Balance Sheet Date. "Balance Sheet Date" shall have the meaning ascribed to it in Section 6.5.3 hereof.

1.20. Basket Amount. "Basket Amount" shall have the meaning ascribed to it in Section 13.2.4 hereof.

1.21. Bundled Sale. "Bundled Sale" shall have the meaning ascribed to it in Section 4.1.3 hereof.

1.22. Bundled Sale Amount. "Bundled Sale Amount" shall have the meaning ascribed to it in Section 4.1.3 hereof.

1.23. Business. "Business" means ARACOR's business and operations as ARACOR currently conducts such business and operations.

1.24. OSI Documents. "OSI Documents" shall have the meaning ascribed to it in Section 8.2 hereof.

-2-

1.25. <u>Change of Control.</u> "Change of Control" shall mean a merger or consolidation of a Person into or with another entity after which the stakeholders of the Person immediately prior to such transaction do not own, immediately following the consummation of the transaction by virtue of their stake in the Person or securities received in exchange for such stake in connection with the transaction, a majority of the voting power of the surviving entity in proportions substantially similar to those that existed immediately prior to such transaction and with substantially the same rights, preferences, privileges and restrictions as the shares they held immediately prior to the transaction, or the sale, transfer or other disposition (but not including a transfer or disposition by pledge or mortgage to a bona fide lender) of all or substantially all of the assets of the Person (other than to a wholly-owned subsidiary). The sale of the assets of the Business or the unit, subdivision, or subsidiary, where the assets of the Business are located, shall be deemed to be a Change of Control.

1.26. Claim Notice. "Claim Notice" shall have the meaning ascribed to it in Section 13.2.1 hereof.

1.27. <u>Claims and Liabilities</u>. "Claims and Liabilities" means, collectively, any losses, claims, liabilities, damages, deficiencies, costs and expenses, including the reasonable fees and expenses of attorneys, accountants and other professionals paid, incurred, accrued or sustained (computed after giving effect to the receipt of any insurance proceeds with respect thereto) by an Indemnitee, net of benefits to the Indemnitee resulting from insurance benefits received.

1.28. <u>Closing</u>. "Closing" means the consummation of the transactions that are contemplated by this Agreement.

1.29. Closing Date. "Closing Date" means the date on which the Closing occurs.

1.30. Code. "Code" means the Internal Revenue Code of 1986, as amended, and the regulations promulgated thereunder.

1.31. <u>Consents and Approvals</u>. "Consents and Approvals" means all authorizations, consents, orders and approvals of federal, state, local and other governmental regulatory bodies, agencies and officials necessary or proper under applicable law for the performance of the parties' obligations under this Agreement and the consummation of the Merger.

1.32. <u>Contingent Consideration</u>. "Contingent Consideration" means the difference of one or more future Contingent Payment(s) and any Other Fees payable as a result of such Contingent Payment(s).

1.33. Contingent Payments. "Contingent Payments" shall have the meaning ascribed to it in Section 4.1 hereof.

1.34. Contingent Payment Dispute Notice. "Contingent Payment Dispute Notice" shall have the meaning ascribed to it in Section 4.7 hereof.

1.35. Contingent Payment Notice. "Contingent Payment Notice" shall have the meaning ascribed to it in Section 4.6.4 hereof.

1.36. Contingent Payment Percentage. "Contingent Payment Percentage" shall have the meaning ascribed to it in Section 4.6.4 hereof.

1.37. CGCL. "CGCL" means the California General Corporation Law, as amended from time to time.

-3-

1.38. <u>Dissenting Shares</u>. "Dissenting Shares" means shares of ARACOR Common Stock which are issued and outstanding immediately prior to the Closing and which are held by a holder who did not vote such shares in favor of the Merger and who has complied with all of the relevant provisions of Section 1300 et seq. of the CGCL.

- 1.39. Earn-Out Period. "Earn-Out Period" shall have the meaning ascribed to it in Section 4.1.1 hereof.
- 1.40. Earn-Out End Date. "Earn-Out End Date" shall have the meaning ascribed to it in Section 4.1.1 hereof.
- 1.41. Effective Time. "Effective Time" shall have the meaning ascribed to it in Section 2.1 hereof.
- 1.42. Environmental Law. "Environmental Law" shall have the meaning ascribed to it in Section 6.25.2 hereof.
- 1.43. ERISA. "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder.
- 1.44. ERISA Affiliate. "ERISA Affiliate" shall have the meaning ascribed to it in Section 6.10.1 hereof.
- 1.45. ERISA Plans. "ERISA Plans" shall have the meaning ascribed to it in Section 6.10.2 hereof.
- 1.46. Existing Leases. "Existing Leases" shall have the meaning ascribed to it in Section 6.11 hereof.
- 1.47. GAAP. "GAAP" shall mean United States generally accepted accounting principles consistently applied.
- 1.48. Hazardous Activity. "Hazardous Activity" shall have the meaning ascribed to it in Section 6.25.1 hereof.
- 1.49. Hazardous Material. "Hazardous Material" shall have the meaning ascribed to it in Section 6.25.1 hereof.
- 1.50. Indemnitee. "Indemnitee" shall have the meaning ascribed to it in Section 13.2.1 hereof.
- 1.51. Indemnitor. "Indemnitor" shall have the meaning ascribed to it in Section 13.2.1 hereof.
- 1.52. Indemnity Cap. "Indemnity Cap" shall have the meaning ascribed to it in Section 13.2.4 hereof.
- 1.53. Individual Documents. "Individual Documents" shall have the meaning ascribed to it in Section 7.1 hereof.
- 1.54. Individual Stock. "Individual Stock" shall have the meaning ascribed to it in Section 7.2 hereof.

-4-

1.55. Initial Payment. "Initial Payment" means an immediate aggregate cash payment equal to Nineteen Million Six Hundred One Thousand Dollars (\$19,601,000).

1.56. Initial Payment Percentage. "Initial Payment Percentage" shall have the meaning ascribed to it in Section 3.3 hereof.

1.57. Leased Properties. "Leased Properties" shall have the meaning ascribed to it in Section 6.11 hereof.

1.58. Lien. "Lien" means any mortgage, pledge, assessment, security interest, lease, sublease, lien, adverse claim, levy, charge, option, right of others or restriction (whether on voting, sale, transfer, disposition or otherwise) or other encumbrance of any kind, whether imposed by agreement, understanding, law or equity, or any conditional sale contract, title retention contract or other contract to give or to refrain from giving any of the foregoing.

1.59. Material Contract. "Material Contract" shall have the meaning ascribed to it in Section 6.13 hereof.

1.60. Merger. "Merger" means the reorganization and merger of OSI Subsidiary into ARACOR as contemplated by this Agreement.

1.61. Merger Consideration. "Merger Consideration" means the sum of the Initial Payment and any Contingent Payment paid.

1.62. Merger Payment. "Merger Payment" means the sum of the Initial Payment and any and all Contingent Payments.

1.63. Net Sales. "Net Sales" shall have the meaning ascribed to it in Section 4.1.2 hereof.

1.64. <u>Other Fees</u>. "Other Fees" means professional advisory fees incurred in connection with the transaction contemplated by this Agreement, payable to Alliant Partners and Gray Cary Ware & Freidenrich LLP in the amounts set forth on <u>Schedule 1.64</u>.

1.65. Pass Through Amounts. "Pass Through Amounts" shall have the meaning ascribed to it in Section 4.1.4 hereof.

1.66. Permits. "Permits" shall have the meaning ascribed to it in Section 6.20 hereof.

1.67. <u>Person</u>. "Person" means any natural person, partnership (whether general or limited), limited liability company, trust, estate, association, corporation, joint venture, firm, custodian, trustee, nominee, or any other individual or entity, including any governmental entity, in its own or representative capacity.

1.68. Personal Property Leases. "Personal Property Leases" shall have the meaning ascribed to it in Section 6.15 hereof.

1.69. Plans. "Plans" shall have the meaning ascribed to it in Section 6.10.1 hereof.

1.70. Products. "Products" shall have the meaning ascribed to it in Section 4.1.4 hereof.

-5-

- 1.71. Proprietary Intellectual Property. "Proprietary Intellectual Property" shall have the meaning ascribed to it in Section 4.1.6 hereof.
- 1.72. SDP. "SDP" shall have the meaning ascribed to it in Section 6.10.3 hereof.
- 1.73. Shareholder Representative. "Shareholder Representative" shall have the meaning ascribed to it in Section 14.1 hereof.
- 1.74. Sick Leave. "Sick Leave" shall have the meaning ascribed to it in Section 9.6 hereof.
- 1.75. Surviving Corporation. "Surviving Corporation" shall have the meaning ascribed to it in Section 2.1 hereof.
- 1.76. Termination Date. "Termination Date" means January 9, 2004.
- 1.77. Third Party Claim. "Third Party Claim" shall have the meaning ascribed to it in Section 13.2.3 hereof.
- 1.78. WARN Act. "WARN Act" shall have the meaning ascribed to it in Section 6.9.3 hereof.

2. THE MERGER AND RELATED MATTERS

2.1. **The Merger**. On the Closing Date, OSI Subsidiary shall be merged with and into ARACOR pursuant to the CGCL, and pursuant to the terms and conditions of this Agreement, and ARACOR shall be the surviving corporation of the Merger (the "<u>Surviving Corporation</u>"). On or before the Closing Date, the parties shall file a Certificate of Merger with the Secretary of State of the State of California and make all other filings or recordings required by law in connection with the Merger. The Merger shall become effective at such time as the Certificate of Merger is duly filed with the Secretary of State of the State of California (the "<u>Effective Time</u>"), at which time each share of the issued and outstanding capital stock of OSI Subsidiary shall convert into one share of the Common Stock of the Surviving Corporation, and such shares shall constitute all the issued and outstanding capital stock of the Surviving Corporation.

2.2. Stock Transfer Books of ARACOR. The stock transfer books of ARACOR shall be closed as of the Closing Date, and no transfer or assignment of ARACOR Common Stock shall take place after the Closing Date, except for the surrender of stock contemplated by Section 2.3 of this Agreement.

2.3. Surrender of ARACOR Stock Certificates. At the Closing, or as soon thereafter as practicable, the ARACOR Shareholders shall surrender to OSI Subsidiary or the Surviving Corporation, as appropriate, each certificate evidencing ARACOR Common Stock in exchange for the right to receive the sum of (i) the immediate cash payment of an amount calculated pursuant to <u>Section 3</u>, which amount shall be set forth with respect to each ARACOR Shareholder on <u>Schedule 3</u>, and (ii) future payments of applicable portions of any Contingent Payments pursuant to <u>Section 4</u>. From and after the Closing, each certificate evidencing ARACOR Common Stock shall be deemed for all purposes to represent and evidence only the right to receive the applicable portion of the Initial Payment pursuant to <u>Section 3</u> and the applicable portion of any Contingent Payments pursuant to <u>Section 4</u>. Except as provided in <u>Section 2.4</u>, unless and until any certificate evidencing ARACOR Common Stock has been surrendered in accordance with this Agreement, the holder of such ARACOR Common Stock shall not have any right to receive any portion of the Merger Consideration. After the Closing, no holder of a certificate evidencing shares of ARACOR Common Stock shall be entitled to receive any dividend or other distribution with regard thereto.

-6-

2.4. Lost, Stolen or Destroyed Certificates. In the event any certificate representing any ARACOR stock shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such certificate to be lost, stolen or destroyed, in a form satisfactory to OSI, and upon verification by OSI to OSI's satisfaction that such certificate had been issued and is outstanding, OSI shall accept such affidavit in lieu of such certificate as if such certificate had been duly surrendered in accordance with Section 2.3. The holder of such lost certificate will be required to indemnify and agree to hold harmless OSI and its transfer agent for any and all damages, liabilities, losses, costs or expenses incurred by OSI or its transfer agent in connection therewith.

3. TREATMENT OF ARACOR COMMON STOCK AND OPTIONS, AND OTHER PAYMENTS AT CLOSING.

3.1. <u>Treatment of ARACOR Common Stock</u>. At the Effective Time, each share of ARACOR Common Stock shall be converted into the right of its owner to receive the applicable Initial Payment Percentage of the Initial Payment net of Other Fees and the right to receive certain Contingent Payments if any. Any shares of ARACOR Common Stock then held by ARACOR shall cease to exist at the Effective Time.

3.2. <u>Treatment of ARACOR Stock Options</u>. At the Effective Time, each ARACOR Stock Option outstanding immediately prior to the Effective Time shall be converted into the right of the ARACOR Optionholder to receive the applicable Initial Payment Percentage of the Initial Payment net of Other Fees and the right to receive certain Contingent Payments if any. All ARACOR Stock Options held by ARACOR Optionholders shall terminate and cease to exist at the Effective Time. No ARACOR Stock Option shall be deemed to be ARACOR Common Stock for any purposes.

3.3. Initial Payment Percentage. The "Initial Payment Percentage" shall indicate a particular ARACOR Shareholder's, or ARACOR Optionholder's, as the case may be, share of the Initial Payment net of Other Fees.

3.3.1. The "Price Per Share" shall be calculated as the following quotient:

A	+	В
С	+	D

Where:

А	=	The Initial Payments
В	=	The aggregate of all exercise prices payable assuming the acceleration and full exercise of all ARACOR Stock Options
С	=	The number of shares of ARACOR Common Stock outstanding as of the Closing
D	=	The number of shares purchasable under all ARACOR Stock Options on the Closing Date, assuming the acceleration and full exercise of all ARACOR Stock Options

3.3.2. With respect to an ARACOR Shareholder, Initial Payment Percentage shall be calculated as a quotient (expressed as a percentage), the numerator of which shall be the number of shares of ARACOR Common Stock held by that ARACOR Shareholder at the Closing, and the denominator of which shall be the sum of (a) the number of shares of ARACOR Common Stock outstanding and (b) the Net Option Shares. Net Option Shares shall be calculated as: D - (B/(Price Per Share)), where D, B and Price Per Share shall have the values ascribed in Section 3.3.1.

-7-

3.3.3. With respect to an ARACOR Optionholder, Initial Payment Percentage shall be calculated as a quotient,

(a) The numerator of which shall be the difference of (i) the number of shares purchasable under all of the ARACOR Optionholder's ARACOR Stock Options, assuming acceleration and full exercise, minus (ii) the quotient of (x) the aggregate exercise price of all of the ARACOR Optionholder's ARACOR Stock Options, assuming acceleration and full exercise, divided by (y) the Price Per Share;

Option Shares.

(b) The denominator of which shall be the sum of (a) the number of shares of ARACOR Common Stock outstanding and (b) the Net

3.4. Payment of Other Fees. The Other Fees shall be paid at the Closing by OSI, as provided in Schedule 1.64.

3.5. **Dissenting Shares**. Notwithstanding anything in this Agreement to the contrary, Dissenting Shares shall not be converted into a right to receive the Merger Payment unless and until the holder or holders thereof shall have failed to perfect or shall have effectively withdrawn or lost their rights to appraisal under the CGCL. The holders of such Dissenting Shares shall be entitled only to such rights as are granted by Section 1300 et seq. of the CGCL. Each holder of Dissenting Shares who becomes entitled to payment for such shares pursuant to Section 1300 et seq. of the CGCL shall receive payment therefore, after the Closing, from the Surviving Corporation in accordance with the CGCL; <u>provided</u>, <u>however</u>, that if any such holder of Dissenting Shares (i) shall have failed to establish such holder's entitlement to relief as a dissenting shareholder as provided in Section 1300 et seq. of the CGCL, (ii) shall have effectively withdrawn such holder's demand for relief as a dissenting shareholder with respect to such Dissenting Shares or lost such holder's right to relief as a dissenting shareholder with respect to such Dissenting Shares or lost such holder's right to relief as a dissenting shareholder with respect to such Dissenting Shares or lost such holder's right to relief as a dissenting shareholder with respect to such Dissenting Shares and each such share shall be converted into the right to receive the Merger Payment, without interest thereon, from Surviving Corporation. Prior to Closing, ARACOR shall give OSI (i) prompt written notice of any demands received by the ARACOR relating to shareholders' rights of appraisal and (ii) the opportunity to direct all negotiations and proceedings with respect to, or settle or offer to settle, any such demands.

3.6. <u>Manner of Distribution of Initial Payment</u>. <u>Schedule "3"</u>, prepared under the assumption that no shares of ARACOR Common Stock shall change ownership prior to the Closing, sets forth (i) each ARACOR Shareholder who shall receive a portion of the Initial Payment at the Closing and subsequent Contingent Payment(s), and for each such ARACOR Shareholder, the number of shares held, the Initial Payment Percentage, the Contingent Payment to be received at the Closing and the method of delivery of the Initial Payment and Contingent Payments for each respective ARACOR Shareholder, all in accordance with the terms of this Agreement and (ii) each ARACOR Optionholder who shall receive a portion of the Initial Payment at the Closing and portions of subsequent Contingent Payments(s), and for each ARACOR Optionholder, the number of options held, the Initial Payment Percentage, the Contingent Payment Percentage, the actual cash amount to be received at the Closing and the method of delivery of the Initial Payment of options held, the Initial Payment Percentage, the Contingent Payment Percentage, the Closing and the method of delivery of the Initial Payment and Contingent Payments for each

-8-

respective ARACOR Optionholder, all in accordance with the terms of this Agreement. <u>Schedule 3</u> sets forth an estimate of amounts to be distributed to each ARACOR Shareholder and ARACOR Optionholder, but shall be amended at the Closing to reflect the actual amount of Other Fees, and the actual cash payment to be received by each ARACOR Shareholder and ARACOR Optionholder at the Closing. If shares of ARACOR Common Stock change ownership prior to the Closing, <u>Schedule 3</u> shall be amended at the Closing to reflect such change. If payment amounts indicated in <u>Schedule 3</u> are inconsistent with the amounts calculated pursuant to this <u>Section 3</u>, the amounts indicated in <u>Schedule 3</u> shall be determinative. The Initial Payment shall be paid as follows:

3.6.1. First, in payment of Other Fees;

3.6.2. Then, to each ARACOR Stockholder and ARACOR Optionholder, the payment to each equal to the product of (a) the applicable Initial Payment Percentage; and (b) the remaining Initial Payment.

4. CONTINGENT PAYMENTS.

4.1. <u>Amount of and Criteria for Contingent Payments</u>. Each of the Contingent Payments described in this <u>Section 4</u> shall be referred to herein as a "<u>Contingent Payment</u>", and collectively as the "<u>Contingent Payments</u>". For each Earn-out Period after the Closing beginning January 2004, OSI shall pay the ARACOR Shareholders and ARACOR Optionholders a Contingent Payment in an amount equal to (a) 7.5% of ARACOR's Net Sales for that Earn-out Period, plus (b) to the extent that such payment does not result in a loss on any particular Pass-Through Amount, 3% of receipts from Pass-Through Amounts, less (c) the Sick Leave Deduction for that Earn-Out Period. Such Contingent Payments shall continue until an aggregate of \$30,000,000 of such Contingent Payments (including any advance payments pursuant to <u>Section 4.4</u> have been made, at which time OSI's obligations under this <u>Section 4</u> shall have been fulfilled and OSI shall be under no further obligation with respect to Contingent Payments. If ARACOR's revenues for calendar 2003 fall short of \$15,000,000, the amount of the shortfall shall be deducted from Net Sales for the first Earn-out Period of 2004 for purposes of calculating the Contingent Payment for that Earn-out Period. The following definitions shall apply to this <u>Section 4</u>:

4.1.1. "Earn-out Period" refers to each of the following: the six-month period from January 1, 2004 through June 30, 2004; each of the years ended June 30, 2005 through and including June 30, 2010; and the six-month period from July 1, 2010 through December 31, 2010 (the "Earn Out Period End Date").

4.1.2. "<u>Net Sales</u>" means cash receipts by OSI and its Affiliates (including Surviving Corporation), but not including any inter-company sales among OSI and its Affiliates (not involving receipts of funds from customers), that are recognized as revenue under GAAP, from:

(a) (i) maintenance or repair of Products, or agreements for the maintenance or repair of Products, (ii) sales of spare parts for and components of Products, and (iii) license fees from the license of Proprietary Intellectual Property; plus

(b) all Bundled Sale Amounts; plus

(c) sale or lease of Products, or the providing of Products on a fee-per-use basis; less

-9-

(d) (i) commissions; (ii) freight and insurance; and (iii) warranty reserves; provided however that except as set forth in <u>Section 4.1.3</u> relating to the possibility of jointly developed products, Net Sales shall not include any products currently marketed by OSI or its Affiliates, or in the future developed solely by OSI and its Affiliates other than the Surviving Corporation.

4.1.3. "Bundled Sale" means a sale by OSI or any OSI Affiliate (including the Surviving Corporation) which is not a sale of a Product but which incorporates Proprietary Intellectual Property (e.g. a jointly developed product). "Bundled Sale Amount" means the amount of receipts derived from a Bundled Sale that are directly attributable to such Proprietary Intellectual Property. For the purposes of determining the Bundled Sale Amount with respect to any particular Bundled Sale, OSI and Individual or his designee shall together make a good faith determination of the percentage of the receipts from the Bundled Sale which shall be considered a Bundled Sale Amount after taking into consideration the relative average sales prices of other products containing the Proprietary Intellectual Property contributed to the Bundled Sale, as compared to those of products containing other intellectual property contributed to the Bundled Sale Amounts shall be resolved in accordance with Section 4.7 hereof and the foregoing factors shall be specifically considered by the arbitrator.

4.1.4. "<u>Pass Through Amounts</u>" means, cash receipts in any earnout period from (a) research and development contracts and grants received; and (b) any construction or civil works projects (e.g. buildings, roadways, parking lots or other similar infrastructure projects) related to Products.

4.1.5. "Products" means (i) products sold by ARACOR as of the Closing, each of which is listed on Schedule 4.1.5(i), or products that are based on such products sold by ARACOR as of the Closing, but that require no more than \$100,000 in enhancements or research and development funding in order to be sold following Closing, each of which is also listed on Schedule 4.1.5(i) and (ii) products under development which have been proposed to outside sources for research and development funding and which receive all necessary research and development funding from sources other than OSI or its Affiliates within 18 months of the Closing, each of which is listed on Schedule 4.1.5(ii).

4.1.6. "Proprietary Intellectual Property" means ARACOR Intellectual Property that is proprietary and the property solely of ARACOR, and that is reduced to documentation or practice as of the Closing.

4.2. Life Insurance Cash-Out. In the event that during the 60 days following the Closing Date, the ARACOR-owned life insurance policy or policies covering Individual has a cash surrender value of less than \$176,000, net of all costs incurred in effecting such cash surrender, the shortfall shall be deducted from the Contingent Payments.

4.3. <u>Sick Leave Deduction</u>. For each Earn-Out Period, OSI shall calculate, as a dollar amount, the amount of Sick Leave taken during such Earn-Out Period by calculating, for each day of Sick Leave taken, the base wages for that day of the employee taking the Sick Leave, then taking the sum of all such dollar amounts so calculated. The Sick Leave Deduction for an Earn-Out Period shall be equal to the sum of all Sick Leave taken calculated pursuant to the preceding sentence, less \$20,000. The Sick Leave Deduction shall be the sole remedy to OSI in connection with the assumption and carry-over of ARACOR's obligations under the sick leave benefit program in effect at ARACOR immediately prior to the Effective Time.

4.4. <u>Time of Contingent Payments</u>. Each payment in fulfillment of a Contingent Payment shall be made within 90 days after the end of the Earnout Period for which the Contingent Payment is due, except to the extent that later payments are required for the final Earn-out Period, which payments shall be made no later than March 31, 2012. If after the Earn-out End Date there remain cash receipts by OSI

-10-

and its Affiliates that would have qualified as Net Sales but for the fact that they have not been recognized as revenues under GAAP, Contingent Payments shall be made with respect to such cash receipts received before the Earn-out End Date, to the extent they later qualify as Net Sales, on or before December 31, 2011. Portions of the Initial Payment shall be considered advanced payment of Contingent Payments as follows: (a) If new orders booked by the Surviving Corporation (as evidenced by firm purchase orders, but not including orders from OSI or its other Affiliates) between the Closing and March 31, 2004 are less than \$6,000,000, but exceed \$5,000,000, then \$1,000,000 of the Initial Payment shall be considered an advance payment of Contingent Payments; and (b) If new orders booked by the Surviving Corporation (as evidenced by firm purchase orders, but not including orders from OSI or its other Affiliates) between the Closing and March 31, 2004 are less than \$6,000,000, then \$1,000,000 of the Initial Payment shall be considered an advance payment of Contingent Payments. If, pursuant to this Section 4.4, any portion of the Initial Payment is considered an advance payment of Contingent Payments. If, pursuant to this Section 4.4, any portion of the Initial Payment is considered an advance payments. In addition, and in accordance with Schedule 6.4, the Contingent Payment for the first Earn-out Period shall be net of any unpaid rent or reimbursable expenses due to the Surviving Corporation from Metara, Inc. during the first Earn-out Period.

4.5. Form of Contingent Payments. All payments made in fulfillment of Contingent Payments shall be made in cash by wire transfer or delivery of bank checks as indicated on <u>Schedule 3</u>, or as otherwise directed by the Shareholder Representative with no less than 10 business days prior written notice.

4.6. Manner of Payment of Contingent Payments. OSI shall distribute payments in fulfillment of Contingent Payments as follows:

4.6.1. First to Alliant Partners calculated in accordance with the formula set forth in Schedule 1.64;

4.6.2. Then to Individual as follows: At any time prior to the thirtieth day prior to the due date for the payment of any Contingent Payment, Individual may give written notice to OSI indicating the amounts paid by Individual in connection with Individual's obligations pursuant to <u>Section 13</u> and the Individual Indemnification Agreement;

4.6.3. Then to the Shareholder Representative as follows: At any time prior to the thirtieth day prior to the due date for the payment of any Contingent Payment, the Shareholder Representative may give written notice to OSI indicating the expenses of any kind advanced by Shareholder Representative in connection with any investigation, audit or proceeding pursuant to Section 4.7 hereof;

4.6.4. Then to the ARACOR Shareholders and ARACOR Optionholders identified in <u>Schedule 3</u>, paying each ARACOR Shareholder or ARACOR Optionholder the amount equal to the product of the remaining amount of the Contingent Payment multiplied by the applicable Contingent Payment Percentage. Each payment shall be accompanied by a statement from the OSI chief financial officer (the "<u>Contingent Payment Notice</u>") setting forth in reasonable detail the calculation of Net Sales and Pass Through Amounts for the applicable Earn Out Period (including an itemization of all constituent elements of Net Sales and Pass Through Amounts. The Shareholder Representative shall simultaneously receive a copy of all such Contingent Payment Notices. "<u>Contingent Payment Percentage</u>" shall be calculated as follows:

(a) With respect to each ARACOR Stockholder, the Contingent Payment Percentage shall be calculated as a quotient the numerator of which is the number of shares of ARACOR Common Stock owned by that ARACOR Stockholder, and the denominator of which is the sum of (i) the number of shares of ARACOR Common Stock outstanding as of the Closing, plus (ii) the aggregate number of shares of ARACOR Common Stock purchasable under the ARACOR Stock Options at the Closing, assuming acceleration and full exercise of all ARACOR Stock Options.

-11-

(b) With respect to each ARACOR Optionholder, the Contingent Payment Percentage shall be calculated as a quotient, the numerator of which is the number of shares of ARACOR Common Stock purchasable under that ARACOR Optionholder's ARACOR Stock Options at the Closing, assuming acceleration and full exercise of those ARACOR Stock Options, and the denominator of which is the sum of (i) the number of shares of ARACOR Common Stock outstanding as of the Closing, plus (ii) the aggregate number of shares of ARACOR Common Stock purchasable under the ARACOR Stock Options at the Closing, assuming acceleration and full exercise of all ARACOR Stock Options.

4.7. Audit; Dispute Resolution. Until the applicable Audit Period Termination Date, OSI shall, upon request, make available during normal business hours all books and records related to the respective Contingent Payments pursuant to this Section 4 for review by the Shareholder Representative. Shareholder Representative may request copies of such books and records, which request shall not be unreasonably denied by OSI. The applicable "Audit Period Termination Date" shall mean 60 days after OSI makes available to the Shareholder Representative financial statements covering the applicable Earn Out Period. In the event that the Shareholder Representative disputes OSI's determination of the amount of the Contingent Payment or whether a payment pursuant to Section 4 should be, or should have been, made, the Stockholder Representative shall notify OSI in writing (a "Contingent Payment Dispute Notice") of the amount, nature and basis of such dispute. The Stockholder Representative and OSI shall first endeavor to resolve such dispute among themselves. If the Stockholder Representative and OSI are unable to resolve the dispute within forty-five (45) calendar days following OSI's receipt of a Contingent Payment Dispute Notice, the dispute shall be submitted to binding arbitration. Such arbitration shall be held in Los Angeles, California and shall be conducted before a single arbitrator in accordance with the Commercial Arbitration Rules of the American Arbitration Association. OSI and the Shareholder Representative shall agree on the arbitrator, provided that if OSI and the Shareholder Representative cannot agree on such arbitrator, either OSI or Shareholder Representative can request that the American Arbitration Association (Los Angeles) select the arbitrator. The arbitrator shall set a limited time period and establish procedures designed to reduce the cost and time for discovery while allowing the parties an opportunity, adequate in the sole judgment of the arbitrator, to discover relevant information from the opposing parties about the subject matter of the dispute. The arbitrator shall rule upon motions to compel or limit discovery and shall have the authority to impose sanctions, including attorneys' fees and costs, to the same extent as a court of competent law or equity, should the arbitrator determine that discovery was sought without substantial justification or that discovery was refused or objected to without substantial justification. The decision of the arbitrator shall be written, shall be in accordance with applicable law and with this Agreement, and shall be supported by written findings of fact and conclusion of law which shall set forth the basis for the decision of the arbitrator. The decision of the arbitrator regarding any issue raised in the Contingent Payment Dispute Notice shall be binding and conclusive. The parties agree to complete such arbitration as expeditiously as reasonably possible.

4.8. <u>Change of Control.</u> In the event that a Change of Control of OSI occurs, or the Surviving Corporation is no longer a subsidiary of OSI, at any time prior to the Earn Out Period End Date and thereafter OSI or the Surviving Corporation breaches <u>Section 4.10</u>, then each ARACOR Shareholder and ARACOR Optionholder shall be entitled to the applicable Contingent Payment Percentage of \$30,000,000 less all Contingent Payments made to date. Such payments shall be made in the manner set forth in <u>Section 4.6</u> hereof.

4.9. <u>Management of Business</u>. Following the Effective Time OSI will provide the Surviving Corporation with financial resources that, in OSI's judgment, are reasonable to support the business of the Surviving Corporation. The parties acknowledge that prior to the Earn Out Period End Date, circumstances may arise under which it would be beneficial to OSI to make changes in the Business (such as, for example the inclusion of other businesses or products under the control and management of ARACOR). In this event, OSI shall reasonably involve Individual, while Individual remains an employee of Surviving Corporation, in planning and executing such changes.

-12-

4.10. Good Faith. In addition to complying with the covenants set forth in this <u>Section 4</u>, OSI agrees to act in good faith to allow ARACOR Shareholders and ARACOR Optionholders to earn the Contingent Payments described in this <u>Section 4</u>.

4.11. <u>No Restriction on Business Management</u>. Nothing in this <u>Section 4</u>, however, shall in any way restrict or otherwise reduce OSI's degree of reasonable discretion in exercising its business judgment in matters pertaining to management of the Surviving Corporation, or to require any act or omission that may have a detrimental effect on OSI or its Affiliates other than the Surviving Corporation.

5. THE CLOSING

5.1. <u>Closing</u>. Provided that all the conditions precedent to the obligations of the parties to this Agreement to consummate the Closing have been satisfied or waived, the Closing shall take place at the offices of OSI at 10:00 a.m., local time, on the Closing Date.

5.2. <u>Closing Deliveries</u>. In order to consummate the Closing, after the parties have confirmed that the Consents and Approvals other than the filing of the Certificate of Merger with the State of California have been given, the parties shall take the actions and make the deliveries described in this <u>Section 5.2</u>.

5.2.1. By ARACOR and Individual. At the Closing, ARACOR and Individual shall deliver, or cause to be delivered, to OSI Subsidiary and OSI all of the following:

(a) ARACOR Common Stock: Share certificates representing no less than 80 percent of ARACOR Common Stock issued and outstanding as of the Effective Time;

(b) *President Certificate*: A certificate of the President of ARACOR, dated as of the Closing Date, certifying the continued accuracy in all material respects of ARACOR's representations and warranties made in this Agreement;

(c) *Board Resolutions*: Unanimous resolutions of ARACOR's Board of Directors approving this Agreement and all transactions contemplated herein;

(d) Action by Written Consent of ARACOR Shareholders: An Action by Written Consent of the ARACOR Shareholders sufficient to approve this Agreement and all transactions contemplated herein;

(e) Opinion of Counsel: An opinion of counsel to ARACOR, dated as of the Closing Date, in the form of Exhibit "5.2.1(e)" attached

hereto;

(f) *Resignations*: Resignations, dated as of the Closing Date, of all officers and directors of ARACOR in form and substance acceptable to OSI and its counsel, including the termination of all employment agreements or other arrangements between each such officer and ARACOR;

(g) Key Person Employment Agreement: The Key Person Employment Agreement between OSI Subsidiary and Individual, in the form attached hereto as Exhibit "5.2.1(g)" duly executed by Individual and dated as of the Closing Date;

(h) *Third Party Consents*. Consents to the transactions contemplated in this Agreement from all third parties from whom such consents are required as set forth in <u>Schedule "6.2";</u>

-13-

(i) Schedule 3: Schedule 3, as described above;

(j) Alliant Letter: a letter from Alliant Partners stating that the Letter Agreement between ARACOR and Alliant Partners dated April 18, 2003 relates only to the transactions contemplated in this Agreement, and approving of <u>Schedule 3</u> to the extent <u>Schedule 3</u> pertains to payments to Alliant Partners;

standing; and

(k) Certificate of Good Standing: A certificate issued by the Secretary of State of the State of California certifying ARACOR's good

(1) Individual Indemnification Agreement: The Indemnification Agreement between OSI, OSI Subsidiary and Individual, in the form attached hereto as Exhibit "5.2.1(1) duly executed by Individual and dated as of the Closing Date.

5.2.2. <u>By OSI Subsidiary and OSI</u>. At the Closing, or at such other time as set forth below, OSI Subsidiary and OSI shall deliver to ARACOR and the ARACOR Shareholders the following:

(a) Board Resolutions: Resolutions of the boards of directors of OSI Subsidiary and OSI approving the Merger;

(b) Opinion of Counsel: An opinion of counsel to OSI Subsidiary and OSI, dated as of the Closing Date, in the form of Exhibit <u>"5.2.2(b)"</u> attached hereto;

(c) Key Person Employment Agreements: Original counterparts of the Key Person Employment Agreements, duly executed by OSI Subsidiary and dated as of the Closing Date; and

(d) *President Certificate*: A certificate of the President of each of OSI and OSI Subsidiary, dated as of the Closing Date, certifying the continued accuracy in all material respects of the representations and warranties made by each of OSI and OSI Subsidiary in this Agreement.

5.3. <u>Closing Agreements</u>. On the Closing Date, the parties shall execute, acknowledge and deliver, or cause to be executed, acknowledged and delivered, such other instruments or documents, in form and substance reasonably acceptable to the other party's counsel, as may be reasonably necessary to consummate the Merger.

6. <u>ARACOR'S REPRESENTATIONS AND WARRANTIES</u>. Subject to and except for the disclosures made, if any, in the ARACOR Disclosure Schedule identified to the particular section number for which the disclosure is made, as a material inducement to OSI Subsidiary and OSI to enter into this Agreement and to consummate the Merger, ARACOR represents and warrants to OSI Subsidiary and OSI as set forth in this <u>Section 6</u>, and that each of such representations and warranties are true as of the date of this Agreement, and shall be true as of the Closing Date. Such representations and warranties shall inure to the benefit of OSI Subsidiary and OSI and each of their successors and assigns. ARACOR acknowledges that OSI Subsidiary and OSI are each entering into this Agreement in reliance upon the truth and accuracy of each such representation and warranty.

6.1. **Due Organization**. ARACOR is a corporation duly organized, validly existing and in good standing under the laws of the State of California, and has full power and authority and all material licenses, permits and franchises to own, operate or lease, as applicable, its assets and to carry on the Business. ARACOR is duly qualified or licensed to do business and is in good standing in all jurisdictions in which the character of the properties it owns or the nature of the business it conducts makes such

-14-

qualification or licensing necessary or proper, except where the failure to so qualify would not, in the aggregate, have a material adverse effect on the Business. ARACOR has made available to OSI Subsidiary complete and correct copies of ARACOR's Articles of Incorporation and By-laws, in each case as amended to the date of this Agreement.

6.2. <u>Power and Authority</u>. ARACOR has all requisite power, authority and capacity to enter into this Agreement and all documents required to be entered into pursuant hereto (this Agreement and such documents being hereinafter referred to as the "<u>ARACOR Documents</u>"), and all corporate action required to be taken by ARACOR to authorize the execution, delivery and performance of this Agreement and all transactions contemplated hereby have been duly and properly taken including, without limitation, all approvals of ARACOR's Board of Directors and, as of the Closing Date, the ARACOR Shareholders. Upon ARACOR's execution and delivery of the ARACOR Documents, assuming the due authorization, execution and delivery by OSI, OSI Subsidiary and Individual, the ARACOR Documents will become valid and binding obligations of ARACOR, enforceable in accordance with their respective terms, except as may be limited by bankruptcy or insolvency laws or similar laws affecting the rights of creditors generally or general principles of equity. The execution and delivery of the ARACOR Documents and the consummation of the transactions contemplated by the ARACOR Documents will not result in the creation of any Lien, charge or encumbrance of any kind or the acceleration of any indebtedness or other material obligation of ARACOR. ARACOR's execution, delivery and consummation of the transactions contemplated by, do not violate or conflict with any provision of, and do not result in a material default under or a material breach of, and are not subject to any consent requirement under, (a) the articles of incorporation or bylaws of ARACOR, as amended, (b) any Material Contract or other material instrument to which ARACOR is a party or by which ARACOR or any of its assets is bound, (c) any order, writ, injunction, decree or judgment of any court or governmental agency, (d) any obligation, law, ordinance, regulation, order or decree applicable to ARACOR or any of its properties; or (e) any permit, concession, franchise or license applicable t

6.3. Capital Structure. The authorized capital stock of ARACOR consists of 10,000,000 shares of ARACOR Common Stock. As of the date of this Agreement, 4,569,798 shares of ARACOR Common Stock were issued and outstanding. Schedule 3 sets forth, among other things, the name of each holder of ARACOR Common Stock, the number of shares held by such ARACOR Shareholder and the address of each such ARACOR Shareholder. Schedule 3 sets forth the name of each ARACOR Optionholder, the number of shares underlying each such stock option, the per share exercise price of each share subject to such option and the address of each such ARACOR Optionholder. No shares of ARACOR Common Stock are held by ARACOR in its treasury, or are reserved for issuance upon the exercise of warrants. No other shares of capital stock or other voting securities of ARACOR are authorized, issued or outstanding. No shares of ARACOR Common Stock are subject to a repurchase option or risk of forfeiture under any restricted stock purchase agreement or other similar agreement. There are no outstanding stock appreciation rights, restricted stock grants or contingent stock grants and there are no other outstanding contractual rights to which ARACOR is a party, the value of which is derived from the value of shares of ARACOR Common Stock. All outstanding shares of ARACOR Common Stock are, and all shares which may be issued upon exercise or conversion of any security issued by ARACOR will be, when issued, duly authorized, validly issued, fully paid and nonassessable and not subject to preemptive rights. As of the date of this Agreement, except as set forth on Schedule 3, there are no bonds, debentures, notes or other indebtedness of ARACOR having the right to vote (or convertible into, or exchangeable for, securities having the right to vote) on any matters on which shareholders of ARACOR may vote. There is no voting trust or other agreement with regard to voting of any share or fractional share of ARACOR Common Stock. Contingent and effective upon the Effective Time, all ARACOR Stock Options shall have been exercised or terminated and, following the Effective Time, no holder of ARACOR Stock Options will have any right to receive shares of common stock in OSI or the Surviving Corporation. Contingent and effective upon the Effective Time, there shall be no outstanding securities, options, warrants, calls, rights, commitments,

-15-

agreements, arrangements or undertakings of any kind to which ARACOR is a party, or by which ARACOR is bound, obligating ARACOR to issue, deliver or sell, or cause to be issued, delivered or sold, shares of capital stock or other voting securities of ARACOR or obligating ARACOR to issue, grant, extend or enter into any such security, option, warrant, call, right, commitment, agreement, arrangement or undertaking.

6.4. Affiliates. ARACOR has no subsidiaries. ARACOR does not own or hold any shares of stock or other voting or ownership interest in any

Person.

6.5. Financial Statements.

6.5.1. ARACOR has delivered to OSI (i) audited financial statements of ARACOR for the 12 months ended December 31 of 2002, 2000 and 1998, (ii) unaudited financial statements of ARACOR for the 12 months ended December 31 of 2001, 1999 and 1997, (iii) unaudited financial statements of ARACOR for the nine months ended September 30, 2003, and (iv) an unaudited balance sheet as of the Balance Sheet Date, as defined below (collectively, the "<u>ARACOR Financial Statements</u>"). The ARACOR Financial Statements were prepared in accordance with generally accepted accounting principles as applied on a consistent basis during the periods covered and fairly present the financial position of ARACOR as of the dates thereof and the results of its operations and cash flows for the periods then ended, subject, in the case of unaudited statements, to the absence of footnotes and to normal year-end audit adjustments which are not material.

6.5.2. Since January 1, 2000, ARACOR has maintained a system of internal accounting controls sufficient to provide reasonable assurance that (a) transactions are executed in accordance with management's general or specific authorizations, (b) transactions are recorded as necessary to permit preparation of financial statements in conformity with generally accepted accounting principles and to maintain asset accountability, (c) access to assets is permitted only in accordance with management's general or specific authorization, and (d) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

6.5.3. Since October 31, 2003 (the "Balance Sheet Date"):

(a) There has not been: (i) any material adverse change in the financial condition, assets, liabilities, personnel or business of ARACOR or in its relationships with its personnel, suppliers, customers, distributors, lessors or others, except changes in the ordinary course of business; (ii) any material damage, destruction or loss, whether or not covered by insurance, adversely affecting the Business or the assets of ARACOR; (iii) any forgiveness or cancellation of debts or claims, or waiver of any rights except in the ordinary course of the Business; (iv) any advances to, loans to, or investments in or transfers of assets to any affiliated Person or entity or any third party; (v) any event or condition of any character materially adversely affecting the Business or the properties of ARACOR; (vi) any material decrease in the actual net worth or in the book value of ARACOR except in the ordinary course of business; (vii) any discharge or satisfaction of any Lien, charge or encumbrance or payment of any liability or obligation, other than current liabilities in the ordinary course of business; (viii) any change in the credit practices of ARACOR or in its methods of maintaining their books, accounts or business records; or (ix) any change in accounting methods, principles or practices by ARACOR affecting its assets, liabilities or business; and

(b) ARACOR has not and will not have (i) declared or paid any dividend or declared or made any distribution on or authorized the creation or issuance of, or issued or otherwise effected any split-up or other reclassification or reorganization of any of its stock or directly or indirectly redeemed, purchased or otherwise acquired any of its outstanding stock except pursuant to repurchase rights in connection with an employee's termination of employment with ARACOR, as set forth in <u>Schedule 6.5.3</u>; (ii) issued or sold or authorized the creation or issuance of any bonds, debentures or other corporate securities

-16-

including without limitation, shares of ARACOR Common Stock except in connection with the exercises of outstanding ARACOR Stock Options; (iii) entered into any transaction other than in the ordinary course of business; (iv) except for this Agreement, entered into any contract of a material nature or (v) with respect to ARACOR's employees or consultants, (A) granted to any employee or consultant any increase in compensation, except as was required under employment agreements in effect as of the Balance Sheet Date or in the ordinary course of business of ARACOR, (B) granted to any employee or consultant any increase in severance or termination pay, except as was required under employment, severance or termination agreements in effect as of the Balance Sheet Date or (C) entered into any employment, severance or termination agreement with any executive officer.

6.6. Liabilities. No corporate debt securities are outstanding, and ARACOR has not entered into any agreement or obligation of any kind and is not subject to any option or right of refusal, to sell all or any part of ARACOR's assets or to merge or consolidate ARACOR with any other entity or concern. Except as reflected in the ARACOR Financial Statements or with respect to any contracts entered into in the ordinary course of business, ARACOR has no material liabilities of any nature, whether absolute accrued, contingent or otherwise; and since the Balance Sheet Date, ARACOR has not incurred any material liabilities other than in the ordinary course of business and on terms and conditions and in amounts consistent with past practices or as otherwise provided on the schedules to this Agreement. As of the Closing, ARACOR shall have no promissory notes outstanding. All expenses and compensation payable to Alliant Partners at the Closing or by reason of this Agreement shall be paid through the Other Fees and Contingent Payments.

6.7. <u>Product Warranties</u>. ARACOR has made available to OSI copies of all written warranties covering ARACOR's products to which ARACOR is a party.

6.8. Taxes. ARACOR has duly and timely filed all federal, state, county and local income, excise, sales, employment and other tax returns and reports required to have been filed by ARACOR up to and including the date hereof, except where the failure to so file would not, in the aggregate, have a material adverse effect on the Business. All such returns are complete and accurate in all material respects and disclose all material taxes required to be paid for in the periods covered thereby. ARACOR has paid all taxes due and payable (whether or not shown on such returns), all assessments, notice of which have been received by ARACOR, and all other taxes, governmental charges, penalties, interest and fines due and payable by ARACOR on or before the date hereof. No audit by a tax authority is pending or threatened with respect to any material taxes due from ARACOR. No issue has been raised by any tax authority in any audit of ARACOR that if raised with respect to any other period not so audited could reasonably be expected to result in a deficiency for any period not so audited. No deficiency or adjustment for any taxes has been proposed, asserted or assessed against ARACOR that has not been resolved or paid or for which an adequate accrual has not been established in accordance with generally accepted accounting principles. ARACOR is not a party to any tax sharing agreements, tax indemnity agreements or similar agreements. As of the Closing Date, there will be no outstanding liability of ARACOR for any due and unpaid taxes of any kind whatsoever, or for interest and penalties on any previously paid or unpaid taxes. ARACOR has neither given nor been requested to give, nor has ARACOR signed, any extension of time or waiver of any statute of limitations with respect to federal, state or local taxes of any kind whatsoever. ARACOR has not received any notice of deficiency or assessment or proposed deficiency or assessment by the Internal Revenue Service or any other taxing authority. There is no pending audit of ARACOR,

6.9. Employee Matters.

6.9.1. (i) ARACOR is not a party to any collective bargaining agreement or other labor union contract, (ii) to ARACOR's knowledge, there are no organizational campaigns, demands or

-17-

proceedings pending or threatened by any labor organization or group of employees seeking recognition or certification as collective bargaining representatives of any group of employees of ARACOR, and (iii) there are no strikes, controversies, slowdowns, work stoppages or material labor disputes, or, to ARACOR's knowledge, threatened involving the employees of ARACOR located at any site of employment of ARACOR.

6.9.2. ARACOR is, and has at all times been, in material compliance with all applicable laws respecting employment and employment practices, terms and conditions of employment, wages, hours of work and occupational safety and health, and to its knowledge is not engaged in any unfair labor practices as defined in the National Labor Relations Act or other applicable law, ordinance or regulation. There are no complaints, controversies, charges, investigations, lawsuits or other proceedings relating to ARACOR pending in any court or with any agency responsible for the enforcement of Federal, state, local or foreign labor or employment laws regarding breach of any express or implied contract of employment, any law or regulation governing employment or the termination thereof or other discriminatory, wrongful or tortious conduct in connection with the employment relationship, which, if adversely determined, would individually or in the aggregate, result in any material liability to ARACOR.

6.9.3. Since January 1, 2002, ARACOR has not effectuated (i) a "plant closing" as defined in the Worker Adjustment and Retraining Notification Act of 1988 (the "<u>WARN Act</u>") affecting any site of employment or one or more facilities or operating units within any site of employment or facility of ARACOR, or (ii) a "mass layoff" (as defined in the WARN Act) affecting any site of employment or facility of ARACOR; nor has ARACOR been affected by any transaction or engaged in layoffs or employment terminations sufficient in number to trigger application of any similar state or local law.

6.9.4. Accumulated Sick Leave (calculated on the basis of employee base wages at the time Sick Leave is taken) does not exceed the equivalent of \$500,000.

6.10. Employee Benefits and ERISA Matters.

6.10.1. <u>Schedule 6.10.1</u> contains a true and complete list of each employment, consulting, bonus, deferred compensation, incentive compensation, stock purchase, stock option, stock appreciation right or other stock-based incentive, severance, change-in-control or termination pay, hospitalization or other medical, disability, life or other insurance, supplemental unemployment benefits, profit-sharing, pension, or retirement plan, program, agreement or arrangement, and each other employee benefit plan, program, agreement or arrangement, sponsored, maintained or contributed to or required to be contributed to by ARACOR, or by any trade or business, whether or not incorporated, that together with ARACOR would be deemed to comprise a controlled group or affiliated service group or be deemed to be under common control or otherwise aggregated for purposes of Sections 414(b), (c), (m) or (o) of the Code (an "<u>ERISA Affiliate</u>"), for the benefit of any current or former employee or director of ARACOR or any ERISA Affiliate (the "<u>Plans</u>").

6.10.2. <u>Schedule 6.10.2</u> identifies each of the Plans that is an "employee welfare benefit plan," or "employee pension benefit plan" as such terms are defined in Sections 3(1) and 3(2) of ERISA (such plans being hereinafter referred to collectively as the "<u>ERISA Plans</u>"). None of the Plans is subject to Title IV of ERISA. Neither of ARACOR nor any ERISA Affiliate has any commitment, whether legally binding or not, to create any additional Plan or modify or change any existing Plan that would affect any current or former employee or director of ARACOR or any ERISA Affiliate other than as required by applicable law, rule or regulation.

6.10.3. With respect to each of the Plans, ARACOR has heretofore delivered or made available or as promptly as practicable after the date hereof shall deliver or make available to OSI Subsidiary true and complete copies of each of the following documents, as applicable: (a) a copy of the

-18-

Plan documents (including all amendments thereto) for each written Plan; (b) a copy of the annual report or Internal Revenue Service Form 5500 Series, if required under ERISA, with respect to each ERISA Plan for the last three Plan years ending prior to the date of this Agreement for which such a report was filed; (c) a copy of the actuarial report, if required under ERISA, with respect to each ERISA Plan for the last three Plan years ending prior to the date of this Agreement for which such a report was filed; (c) a copy of the actuarial report, if required under ERISA, with respect to each ERISA Plan for the last Plan year ending prior to the date of this Agreement; (d) a copy of the most recent Summary Plan Description ("SPD"), together with all Summaries of Material Modification issued with respect to such SPD, if required under ERISA, with respect to each ERISA Plan, and all other material employee communications relating to each ERISA Plan; (e) if the Plan is funded through a trust or any other funding vehicle, a copy of the trust or other funding agreement (including all amendments thereto) and the latest financial statements thereof, if any; (f) all contracts relating to the Plans with respect to which ARACOR or any ERISA Affiliate may have any material liability, including insurance contracts, investment management agreements, subscription and participation agreements and record keeping agreements; and (g) the most recent determination letter (or opinion letter, if applicable) received from the Internal Revenue Service with respect to each Plan that is intended to be qualified under Section 401(a) of the Code.

6.10.4. Neither ARACOR, any ERISA Affiliate, any of the ERISA Plans, any trust created thereunder, nor to ARACOR's knowledge, any trustee or administrator thereof has engaged in a transaction or has taken or failed to take any action in connection with which ARACOR, any ERISA Affiliate, any ERISA Plan or any such trust could be subject to any material liability for either a civil penalty assessed pursuant to Section 502(i) of ERISA or a tax imposed pursuant to Section 4975(a) or (b), 4976 or 4980B of the Code.

6.10.5. All contributions which ARACOR or any ERISA Affiliate is required to pay under the terms of each of the ERISA Plans have been timely paid in full or properly recorded on the financial statements or records of ARACOR.

6.10.6. Each of the Plans has been operated and administered in all material respects in accordance with applicable laws, including but not limited to ERISA and the Code.

6.10.7. There has been no material failure by any of the ERISA Plans that is intended to be "qualified" within the meaning of Section 401(a) of the Code to meet the requirements of such qualification. ARACOR has applied for and received a currently effective determination letter, or, if applicable, has received an opinion letter, from the Internal Revenue Service stating that it is so qualified, and to ARACOR's knowledge no event has occurred since the issuance of such letter that could reasonably be expected to cause the loss of such qualified status, or ARACOR has time remaining to apply under applicable Treasury Regulations or IRS pronouncements for such determination or opinion letter.

6.10.8. No Plan provides benefits, including without limitation death or medical benefits (whether or not insured), with respect to current or former employees of ARACOR or any ERISA Affiliate after retirement or other termination of service (other than (i) coverage mandated by applicable laws, (ii) death benefits or retirement benefits under any "employee pension plan," as that term is defined in Section 3(2) of ERISA, or (iii) benefits, the full direct cost of which is borne by the current or former employee (or beneficiary thereof)).

6.10.9. The consummation of the transactions contemplated by this Agreement will not (i) entitle any current or former employee, officer or director of ARACOR or any ERISA Affiliate to severance pay, or any other similar termination payment, or (ii) accelerate the time of payment or vesting, or increase the amount of or otherwise enhance any benefit due, or any option or other equity security held by, any such employee, officer or director.

-19-

6.10.10. There are no pending or, to ARACOR's knowledge, threatened claims by or on behalf of any Plan, by any employee or beneficiary under any such Plan or otherwise involving any such Plan (other than routine claims for benefits).

6.10.11. ARACOR has not made any payments, is not obligated to make any payments, nor is it a party to any contract, agreement or other arrangement which could result in the payment by ARACOR of an "Excess Parachute Payment" as that term is used in Section 280G of the Code.

6.10.12. Since the Balance Sheet Date, ARACOR has not adopted any Plan.

6.11. Real Property. ARACOR does not own any real property. Schedule 6.11 sets forth a true and correct list of all real properties currently leased by ARACOR ("Leased Properties"), in each case, setting forth (a) the lessor and lessee thereof and the date and term of each of the leases, licenses or other agreements (the "Existing Leases") for the Leased Properties, and (b) street address of each property covered thereby, and (c) a brief description of the principal use of such property by ARACOR. The Existing Leases for the Leased Properties are valid, binding and in full force and effect, except as may be limited by bankruptcy or insolvency laws or similar laws affecting the rights of creditors generally or general principles of equity, and have not been amended, and neither ARACOR, nor to the knowledge of ARACOR any other party thereto is in default or breach under any such Existing Lease, except for such non-monetary defaults as have not had and are not reasonably likely to have, individually and in the aggregate, a material adverse effect on ARACOR. To ARACOR's knowledge, no event has occurred which, with the passage of time or the giving of notice or both would cause a breach of or default under any of such Existing Leases. The interests of ARACOR in the Existing Leases for the Leased Properties are free and clear of all Liens against, or resulting from the actions of ARACOR. The improvements constructed at each facility subject to an Existing Lease for Leased Properties, including all leasehold improvements, building systems, equipment and all fixtures owned, leased or used by ARACOR at such facilities are to ARACOR's knowledge: (a) in good operating condition and repair in all material respects, subject to ordinary wear and tear; (b) not in need of material repair or correction except for ordinary routine maintenance and repair; (c) sufficient in all material respects for the operation of the Business; and (d) structurally sound in all material respects with no material defects. All utilities presently serving the Leased Properties are adequate to service the Business. With respect to each such Leased Properties: (a) ARACOR has a valid leasehold interest in the Leased Properties; (b) ARACOR has not received notice of any special assessment for which ARACOR is liable that may encumber any of the Leased Properties, except for any real property or taxes passed through to ARACOR pursuant to the terms of the Existing Leases and are reflected in the ARACOR Financial Statements; (c) ARACOR has not received notice of any governmental decree or order for the whole or any portion of the Leased Properties or any property providing access thereto to be sold or condemned, expropriated or otherwise taken by any public authority with or without payment of compensation therefor, nor to the knowledge of ARACOR has any such condemnation, expropriation or taking been proposed by any governmental body.

6.12. Other Assets. ARACOR is the sole and exclusive owner of all right, title and interest in, and has good and indefeasible title to, all the Assets. Schedule 6.12 hereto sets forth a true and complete list of all capital equipment owned by ARACOR with a book value in excess of \$10,000.

6.13. <u>Material Contracts</u>. <u>Schedule 6.13</u> hereto sets forth a complete and accurate list of all material contracts, agreements, options and other commitments, oral or written, to which ARACOR is a party or by which ARACOR or the Business is bound including, without limitation, (a) any non-competition agreement or any other agreement or obligation which purports to limit in any material respect the manner in which, or the localities in which, all or any material portion of the Business is conducted or, assuming the completion of the Merger, the business of the Surviving Corporation would be conducted, (b) any exclusive supply or purchase contracts or any exclusive requirements contracts or (c) any contract or other agreement

-20-

which would prohibit or materially delay the consummation of the Merger or any of the transactions contemplated by this Agreement (collectively, the "<u>Material</u> <u>Contracts</u>"). ARACOR has made available to OSI accurate and complete copies of all the Material Contracts. Each of the Material Contracts is in full force and effect and to ARACOR's knowledge there is no existing condition, event, including, without limitation, the consummation of the Merger, that presently exists or with notice or the passage of time would give any third party any right to terminate or materially adversely affect ARACOR's rights thereunder. To ARACOR's knowledge, no party to any Material Contract is in breach of any provision of, in violation of, or in default under the terms of any Material Contract.

6.14. <u>Related Party Transactions</u>. To ARACOR's knowledge, no officer, director or shareholder of ARACOR has or has had, directly or indirectly, (i) an economic interest in any entity which furnished or sold, or furnishes or sells, services or products that ARACOR furnishes or sells, or proposes to furnish or sell, (ii) an economic interest in any entity that purchases from or sells or furnishes to, ARACOR, any goods or services or (iii) a beneficial interest in any Material Contract; <u>provided</u>, that ownership of no more than one percent (1%) of the outstanding voting stock of a publicly traded corporation shall not be deemed an "economic interest in any entity" for purposes of this <u>Section 6.14</u>.

6.15. <u>Personal Property Leases</u>. <u>Schedule 6.15</u> hereto sets forth a complete and accurate list of all material leases of personal property (the "<u>Personal Property Leases</u>") that ARACOR uses in the Business.

6.15.1. The Personal Property Leases are in full force and effect and are valid, binding, and enforceable in accordance with their respective terms except as may be limited by bankruptcy or insolvency laws or similar laws affecting the rights of creditors generally or general principles of equity;

6.15.2. ARACOR and to ARACOR's knowledge each other party to a Personal Property Lease has complied with all material commitments and obligations on its part to be performed or observed under each Personal Property Lease;

6.15.3. ARACOR has not received any notice of a default, offset or counterclaim under any Personal Property Lease, or any other communication calling upon ARACOR to comply with any provision of any Personal Property Lease or asserting noncompliance, and no event or condition has happened or presently exists that constitutes a default or, after notice or lapse of time or both, would constitute a default under, any Personal Property Lease; and

6.15.4. There is no security interest, Lien, encumbrance or claim on ARACOR's leasehold interest created under any of the Personal Property Leases.

6.16. <u>Insurance</u>. <u>Schedule 6.16</u> sets forth a complete list of all of the insurance policies held by ARACOR that insure any assets of ARACOR. ARACOR maintains insurance coverage in such amounts and covering such risks as are in accordance with normal industry practice for companies engaged in business similar to that of ARACOR.

6.17. Intellectual Property.

6.17.1. <u>Schedule 6.17.1</u> contains a complete and accurate list of all patents, trademarks, trade names, service marks, brand names, and copyrights, and any applications for any of the foregoing, owned by ARACOR, and a complete and correct list of all material licenses and other agreements relating to any of the foregoing. To ARACOR's knowledge, ARACOR owns or has a valid right to use all registered trademarks, service marks, trade names, Internet domain names, designs, slogans, together with all applications, registrations and goodwill related to the foregoing (collectively, the "<u>ARACOR</u> <u>Trademarks</u>"); issued patents (including any registration, continuations, continuations-in-part, renewals and applications for

-21-

any of the foregoing); registered copyrights (including any registrations, renewals and applications for any of the foregoing); ARACOR Software (as defined below); trade secrets and other confidential information, know-how, proprietary processes, product designs, formulae, algorithms, models, and methodologies (collectively, the "<u>ARACOR Trade Secrets</u>") necessary for the conduct of the Business (all such intellectual property being referred to herein as the "<u>ARACOR Intellectual Property</u>"). For purposes of this <u>Section 6.17</u>, "<u>ARACOR Software</u>" means any and all of the following items that are necessary to the conduct of the Business (a) computer programs, including any and all software implementation of algorithms, models and methodologies, whether in source code or object code, (b) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (c) descriptions, flowcharts and other work product used to design, plan, organize and develop any of the foregoing, (d) all documentation, including user manuals and training materials, relating to any of the foregoing.

6.17.2. Except with respect to any licenses granted to third parties, all of which are listed on <u>Schedule 6.17.2</u>, ARACOR Intellectual Property owned by ARACOR is free and clear of all Liens.

6.17.3. To ARACOR's knowledge, all ARACOR Intellectual Property owned by ARACOR that is necessary for the conduct of the Business is valid and subsisting, in full force and effect, and has not been canceled, has not expired, nor has it been abandoned. There is not pending or, to ARACOR's knowledge, threatened opposition, interference or cancellation proceeding before any court or registration authority in any jurisdiction against any registrations in respect of ARACOR Intellectual Property (other than ARACOR Trademarks) owned by ARACOR. All past and present ARACOR employees and consultants have assigned to ARACOR all material intellectual property created in the scope of their employment or engagement by ARACOR.

6.17.4. To ARACOR's knowledge, the conduct of the Business does not infringe upon (either directly or indirectly such as through contributory infringement or inducement to infringe) any intellectual property rights owned or controlled by any third party. There are no claims or suits pending and as to which ARACOR has received actual notice nor, to the knowledge of ARACOR, are any claims or suits threatened, and ARACOR has not received any notice of a third-party claim or suit, (a) alleging that its activities or the conduct of its Business infringes upon, violates, or constitutes the unauthorized use of the intellectual property rights of any third party or (b) challenging the ownership, use, validity or enforceability of any ARACOR Intellectual Property, which in any case would have a material adverse effect.

6.17.5. There are no written settlements, forbearance to sue, consents, judgments, or orders or similar obligations which in any material respect (a) restrict the right of ARACOR to use any ARACOR Intellectual Property owned by ARACOR, or (b) restrict the Business in order to avoid infringing upon a third party's intellectual property rights. There are no agreements that permit third parties to use any ARACOR Intellectual Property owned by ARACOR.

6.17.6. ARACOR takes reasonable measures to protect the confidentiality of ARACOR Trade Secrets, including (a) requiring its employees and independent contractors having access thereto to execute written agreements that protect the confidentiality of ARACOR Trade Secrets and (b) requiring all licensees to maintain the confidentiality of ARACOR Trade Secrets. To the knowledge of ARACOR, no ARACOR Trade Secret has been knowingly disclosed or authorized to be disclosed to any third party other than pursuant to a nondisclosure agreement or other appropriate instrument that protects ARACOR's proprietary interests in and to such ARACOR Trade Secrets. To the knowledge of ARACOR, no party to any nondisclosure agreement or nondisclosure obligation relating to ARACOR Trade Secrets is in breach or default thereof.

6.17.7. To the knowledge of ARACOR, no third party is misappropriating, infringing, or violating any ARACOR Intellectual Property owned by ARACOR other than immaterial disputes concerning use by a third party of Trademarks of ARACOR.

-22-

6.17.8. The consummation of the Merger and the other transactions contemplated by this Agreement shall not result in the loss or impairment of ARACOR's right to own or use any of ARACOR Intellectual Property, and will not require the consent of any governmental authority to transfer rights in or ownership of any ARACOR Intellectual Property owned by ARACOR.

6.18. <u>Corporate Records</u>. The minute books of ARACOR have been provided to OSI and contain in all material respects true and complete minutes and records of all formal meetings and proceedings and other formal action of ARACOR's shareholders and directors, and all meetings and other corporate proceedings of ARACOR's shareholders and directors have been duly, legally and properly held or taken. The stock certificate books of ARACOR have been provided to OSI and all the signatures that appear on the certificates that represent shares of stock of ARACOR (including all certificates that are now outstanding and all certificates that have heretofore be canceled) are the true signatures that they purport to be and were actually affixed to the respective documents by the persons whose names they represent.

6.19. <u>Claims and Litigation</u>. There is no claim, legal proceeding, action, suit, investigation or other proceeding pending, or to ARACOR's knowledge threatened, against ARACOR that could adversely affect the Business or the transactions contemplated by this Agreement. ARACOR is not subject to any judgment, order or decree entered in any lawsuit or proceeding which could have a material adverse effect on ARACOR.

6.20. Licenses and Permits. Each of ARACOR and, to ARACOR's knowledge, its officers, directors and other personnel hold, to the extent legally required, all federal, state, local and foreign governmental approvals, authorizations, certificates, filings, franchises, orders, registrations, findings of suitability, licenses, notices, permits, applications and rights, including all authorizations under any applicable law ("Permits"), necessary for ARACOR to own, lease or operate its properties and assets and to carry on the Business except to the extent that the failure to hold such Permits would not in the aggregate have a material adverse affect on ARACOR, and there has occurred no material default under any such Permit. To the knowledge of ARACOR, no event has occurred, and no facts or circumstances exist, which would prevent ARACOR from obtaining all necessary material Permits to allow ARACOR to conduct the Business.

6.21. <u>Compliance with Law</u>. ARACOR and, to ARACOR's knowledge, each of its officers, directors and other key personnel, is in compliance with all applicable laws with respect to the Business.

6.22. <u>Condition of Assets</u>. None of the Assets, or any individual item of the Assets fails or fail to conform in any material respect to any applicable law, ordinance or regulation. The Assets are free and clear of all Liens, encumbrances, pledges or charges, except as reflected in the Financial Statements or in <u>Schedule 6.22</u> and except for Liens for taxes not yet due and payable and such imperfections of title and encumbrances, if any, which are not material in character, amount or extent, and which do not materially detract from the value, or materially interfere with the present use, of the property subject thereto or affected thereby. All Liens reflected in <u>Schedule 6.22</u> shall have been terminated as of the Closing. All tangible Assets are in good condition and working order, excepting ordinary wear and tear.

6.23. <u>Shareholders' Claim</u>. As of the Closing, no ARACOR Shareholder has any claim against ARACOR or the Assets of any kind or nature with respect to such ARACOR Shareholder's status as a shareholder, except claims, if any, arising from this Agreement and the documents delivered pursuant to this Agreement.

6.24. <u>Contributions</u>. ARACOR has not established or maintained any intentionally unrecorded fund or asset for or on behalf of ARACOR for any purpose, and has not made any intentionally false or artificial entry on any of its books or records for or on behalf of ARACOR for any reason.

-23-

6.25. Environmental Matters.

6.25.1. Except as would not be reasonably expected to result in a material adverse effect on ARACOR, during the five years preceding the Effective Time, (a) ARACOR has not caused any Hazardous Materials to be present on or in the facilities of the Business or at any geologically or hierologically adjoining property, including any Hazardous Materials (as defined below) contained in barrels, above or underground storage tanks, landfills, land deposits, dumps, equipment (whether moveable or fixed) or other containers, either temporary or permanent, and deposited or located in land, water, sumps, or any other part of said facilities or such adjoining property, or incorporated into any structure therein or thereon; and (b) ARACOR has not permitted or conducted any Hazardous Activity (as defined below) that was not in compliance with all applicable Environmental Laws with respect to the Business or its facilities or any other properties or assets (whether real, personal, or mixed) that relate to the operations of ARACOR. As used in this Agreement, "Hazardous Activity," means the distribution, generation, handling, importing, management, manufacturing, processing, production, refinement, release, storage, transfer, transportation, treatment, or use (including any withdrawal or other use of groundwater) of Hazardous Materials in, on, under, about, or from a facility or any part thereof into the environment; and "Hazardous Materials" means any waste or other substance that is listed, defined, designated, or classified as, or otherwise determined to be hazardous, radioactive, or toxic or a pollutant or a contaminant under or pursuant to any environmental law, including any admixture or solution thereof, and specifically including petroleum and all derivatives thereof or synthetic substitutes therefor and asbestos or asbestos-containing materials.

6.25.2. Seller's operations are, in all material respects, in compliance with all applicable federal, state, and local laws pertaining to hazardous substances, wastes, discharges, emissions, disposal, dumping, burial or other forms of pollution, as regulated by federal, state or local law. There are no legal, administrative, arbitral or other proceedings, claims, actions, causes of action, private environmental investigations or remediation activities or governmental investigations of any nature pending or to ARACOR's knowledge threatened against ARACOR seeking to impose, or that could reasonably be expected to result in the imposition of, on ARACOR any liability or obligation arising under common law or under any Environmental Law (as defined below), which liability or obligation could reasonably be expected to have a material adverse effect on ARACOR. To the knowledge of ARACOR, there is no reasonable basis for any such proceeding, claim, action or governmental investigation relating to alleged Hazardous Activity that would impose any liability or obligation that could reasonably be expected to have a material adverse effect on ARACOR. "Environmental Law" means any federal, international, state, local or other law, statute, rule, regulation, ordinance or treaty relating to any Hazardous Materials or to the protection of the environment generally, including, without limitation, (a) the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended by the Superfund Amendments and Reauthorization Act (42 U.S.C. Section 6901, et seq.), (b) the Federal Water Pollution Control Act (33 U.S.C. Section 1251, et seq.), (c) the Clean Air Act (42 U.S.C. Section 6902, et seq.), (e) the Toxic Substances Control Act (15 U.S.C. Section 2601-2629), (f) the Hazardous Materials Transportation Act (49 U.S.C. Section 1801, et seq.), (e) any of the regulations adopted or publications promulgated pursuant to any the foregoing.

6.26. <u>Brokers or Finder's Fees</u>. Except as provided in the ARACOR Disclosure Schedule, no agent, broker, Person or firm acting on behalf of ARACOR is, or will be, entitled to any investment banking or broker's or finder's fees from any of the parties hereto, or from any Person controlling, controlled by, or under common control with any of the parties hereto, in connection with this Agreement or any of the transactions contemplated hereby. ARACOR has made available to OSI true and correct copies of all agreements between brokers or finders and ARACOR in effect as of the Closing.

-24-

6.27. **Full Disclosure**. The representation and warranties of ARACOR contained in this Agreement, and each exhibit, schedule (including, without limitation, the ARACOR Disclosure Schedule), certificate delivered by ARACOR pursuant to this Agreement are accurate, correct and complete in all material respects as of the date delivered, and do not contain any untrue statement of a material fact or, omit to state a material fact necessary in order to make the statements and information contained therein, in light of the circumstances under which they were made, not misleading. There is no fact, presently known, that materially adversely affects the ability of ARACOR to perform fully under this Agreement or to consummate the Merger that has not been set forth or described in this Agreement or in the ARACOR Disclosure Schedule.

7. **INDIVIDUAL'S REPRESENTATIONS AND WARRANTIES**. As a material inducement to OSI Subsidiary and OSI to enter into this Agreement and to consummate the Merger, Individual represents and warrants to OSI Subsidiary and OSI as set forth in this <u>Section 7</u>. Each of such representations and warranties are true as of the date of this Agreement, and shall be true as of the Closing Date. Each such representation and warranty shall inure to the benefit of OSI Subsidiary and OSI and each of their successors and assigns. Individual acknowledges that OSI Subsidiary and OSI are each entering into this Agreement in reliance upon the truth and accuracy of each representation and warranty provided by Individual. Individual represents and warrants that:

7.1. **Power and Authority**. Individual has all requisite power, authority and capacity to enter into this Agreement and all documents required to be entered into pursuant hereto to which such Individual is a party (the "<u>Individual Documents</u>"), and any action required to be taken by Individual to authorize the execution, delivery and performance of this Agreement and all transactions contemplated hereby have been duly and properly taken. Upon Individual's execution and delivery of the Individual Documents, assuming the due authorization, execution and delivery by ARACOR, OSI and OSI Subsidiary, the Individual Documents will become valid and binding obligations of Individual, enforceable in accordance with their respective terms, except as may be limited by bankruptcy or insolvency laws or laws affecting the rights of creditors generally or general principles of equity. Individual's execution and delivery of the Individual Documents and his consummation of the transactions contemplated by the Individual Documents are not prohibited by, do not violate or conflict with any provision of, and do not result in a default under or a breach of (a) any material contract, agreement or other instrument to which Individual is a party or by which such Individual or any of his or its respective assets is bound, (b) any order, writ, injunction, decree or judgment of any court or governmental agency, (c) any obligation, law, ordinance, regulation, order or decree applicable to such Individual or any of their properties, or (d) the operating agreement or other governing documents of Individual (if any).

7.2. **Individual Stock**. The ARACOR Common Stock held by such Individual (the "<u>Individual Stock</u>") is free and clear of any security interest, Lien, pledge, encumbrance, claim, charge, agreement, right, option, warrant and restriction of any kind, nature or description. No option, warrant, restriction, right or other agreement or commitment is currently outstanding or existing with respect to the Individual Stock. Such Individual is, and immediately prior to the time the Closing occurs, will continue to be, the record and beneficial owner of the Individual Stock, free and clear of any security interest, Lien, pledge, encumbrance, claim, charge, agreement, right, option, warrant or restriction of any kind, nature or description, and no other Person owns or has any interest in or to any share of stock of ARACOR or any fractional share, including, without limitation, any voting right, and there is no voting trust with regard to any share of stock of ARACOR or any fractional share.

8. <u>CARDINAL'S AND OSI SUBSIDIARY'S REPRESENTATIONS AND WARRANTIES</u>. Subject to and except for any disclosure made in this Agreement and to information contained in the OSI Disclosure Schedule that is specifically referenced to a paragraph number of this <u>Section 8</u>, as a material inducement to ARACOR and Individual to enter into this Agreement and to

-25-

consummate the Merger, each of OSI and OSI Subsidiary jointly and severally represents and warrants to ARACOR and Individual as set forth in this <u>Section 8</u>, and that each of such representations and warranties are true as of the date of this Agreement, and shall be true as of the Closing Date, and shall inure to the benefit of ARACOR and Individual and each of their successors and assigns. OSI and OSI Subsidiary each acknowledge that ARACOR and Individual are each entering into this Agreement in reliance upon the truth and accuracy of each such representation and warranty.

8.1. **Due Organization**. Each of OSI and OSI Subsidiary is a corporation duly organized, validly existing and in good standing under the laws of the State of California and has full power and authority and all requisite licenses, permits and franchises to own, operate or lease, as applicable, its assets and to carry on their business. Each of OSI and OSI Subsidiary is duly qualified or licensed to do business and is in good standing in all jurisdictions in which the character of the properties it owns or the nature of the business it conducts makes such qualification or licensing necessary or proper, except where the failure to so qualify would not, in the aggregate, have a material adverse effect on OSI's or OSI Subsidiary's business or assets.

8.2. **Power and Authority**. OSI and OSI Subsidiary each have all requisite power, authority and capacity to enter into this Agreement and all documents required to be entered into pursuant hereto (this Agreement and such documents being hereinafter referred to as the "<u>OSI Documents</u>"), and all corporate and other action required to be taken by OSI and OSI Subsidiary to authorize the execution, delivery and performance of this Agreement and all transactions contemplated hereby have been duly and properly taken including, without limitation, all approvals of OSI's and OSI Subsidiary's Board of Directors. Upon OSI's and OSI Subsidiary's execution and delivery of the OSI Documents, assuming the due authorization, execution and delivery by ARACOR and Individual, the OSI Documents will become valid and binding obligations of OSI and OSI Subsidiary, as applicable, enforceable in accordance with their respective terms, except as may be limited by bankruptcy or insolvency laws or similar laws affecting the rights of creditors generally or general principles of equity. The execution and delivery of the OSI Documents and the consummation of the transactions contemplated by the OSI Documents will not result in the creation of any Lien, charge or encumbrance of any kind or the acceleration of any indebtedness or other obligation of OSI or OSI Subsidiary. OSI and OSI Subsidiary's execution, delivery and consummation of the transactions contemplated in the OSI Documents are not prohibited by, do not violate or conflict with any provision of, and do not result in a default under or a breach of (a) the articles of incorporation or bylaws of OSI or OSI Subsidiary, (b) any material contract, agreement or other instrument to which OSI or OSI Subsidiary is a party or by which OSI or OSI Subsidiary or any of their respective assets is bound, (d) any order, writ, injunction, decree or judgment of any court or governmental agency, or (e) any obligation, law, ordinance, regulation, order or decree applicable to OSI or OSI Subsidiary o

8.3. <u>SEC Filings</u>. During its current fiscal year and the three preceding fiscal years, OSI has timely filed with the Securities and Exchange Commission all filings required under applicable statutes, rules and regulations, except for such filings the absence of which, or the lateness of which, would not have a material adverse effect on OSI.

8.4. **Full Disclosure**. The representations and warranties of OSI contained in this Agreement, and each exhibit, schedule (including, without limitation, the OSI Disclosure Schedule), certificate delivered by OSI pursuant to this Agreement are accurate, correct and complete in all material respects as of the date delivered, and do not contain any untrue statement of a material fact or, omit to state a material fact necessary in order to make the statements and information contained therein, in light of the circumstances under which they were made, not misleading. There is no fact, presently known, that materially adversely affects the ability of OSI to perform fully under this Agreement or to consummate the Merger that has not been set forth or described in this Agreement or in the OSI Disclosure Schedule.

-26-

8.5. No Significant Subsidiary. The transactions contemplated by this Agreement do not require a Form 8-K to be filed with the Securities and Exchange Commission with which financial statements are required in accordance with Regulation S-X.

8.6. **Brokers or Finder's Fees**. No agent, broker, Person or firm acting on behalf of OSI is, or will be, entitled to any investment banking or broker's or finder's fees from any of the parties hereto, or from any Person controlling, controlled by, or under common control with any of the parties hereto, in connection with this Agreement or any of the transactions contemplated hereby.

9. COVENANTS OF THE PARTIES

9.1. <u>Cooperation</u>. Upon the terms and subject to the conditions set forth in this Agreement, each of the parties agrees to use all commercially reasonable efforts to take, or cause to be taken (including through its officers and directors and other appropriate personnel), all actions, and to do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the Merger and the other transactions contemplated by this Agreement, including (a) the obtaining of all necessary actions or nonactions, waivers, Consents and Approvals from governmental agencies and the making of all necessary registrations and filings (including filings with governmental agencies, if any) and the taking of all necessary consents, approvals or waivers from third parties, (c) the defending of any lawsuits or other legal proceedings, whether judicial or administrative, challenging this Agreement or the consummation of any of the transactions contemplated herein, including seeking to have any stay or temporary restraining order entered by any court or other governmental agency vacated or reversed and (d) the execution and delivery of any additional instruments necessary to consummate the transactions contemplated by, and to fully carry out the purposes of, this Agreement; provided, however, that OSI shall not be required to take any actions in connection with, or agree to, any hold separate order, sale, divestiture, or disposition of any plants, assets and businesses of OSI or any of its subsidiaries or of ARACOR.

9.2. <u>State Statutes</u>. In connection with and without limiting the foregoing, ARACOR, OSI and the OSI Subsidiary shall (including through its officers and directors and other appropriate personnel) (i) take all commercially reasonable action necessary to ensure that no U.S. state takeover, business combination, control share, fair price or fair value statute or similar statute or regulation is or becomes applicable to the Merger, this Agreement or any of the transactions contemplated hereby or (ii) if any U.S. state takeover, business combination, control share, fair price or fair value statute or regulation becomes applicable to the Merger, this Agreement or any other transaction contemplated hereby, take all commercially reasonable action to ensure that the Merger and the other transactions contemplated hereby may be consummated as promptly as practicable on the terms contemplated and otherwise to minimize the effect of such statute or regulation on the Merger, this Agreement and the other transactions contemplated hereby

9.3. Conduct of Business.

9.3.1. Prior to the Closing, ARACOR shall carry on the Business in all material respects in the ordinary course of business and use all commercially reasonable efforts to preserve intact its current business organizations, keep available the services of its current officers and key employees, maintain in effect all material federal, state and local licenses, approvals and authorizations, including, without limitation, all material licenses, permits, qualifications and authorizations to do business in each jurisdiction in which it is so licensed, qualified or authorized, comply with the terms of all material contracts and preserve its relationships consistent with past practice with customers, suppliers, licensors, licensees, distributors, employees and others having business dealings with them to the end that their goodwill and ongoing businesses shall be unimpaired in all material respects at the Closing.

-27-

9.3.2. Except as expressly permitted or contemplated by the terms of this Agreement, without limiting the generality of the foregoing, prior to the Closing, ARACOR shall not, without the prior written consent of OSI:

(a) (1) declare, set aside or pay any dividends on, or make any other distributions in respect of, any of its capital stock; (2) split, combine or reclassify any of its capital stock or issue or authorize the issuance of any other securities in respect of, in lieu of or in substitution for shares of its capital stock other than as a result of the exercise of ARACOR Stock Options; or (3) purchase, redeem or otherwise acquire or amend the terms of any shares of capital stock of ARACOR or any other securities thereof or any rights, warrants, options to acquire or any securities convertible into or exchangeable for any such shares or other securities.

(b) authorize for issuance, issue, deliver, sell, grant, pledge or otherwise encumber or amend any shares of its capital stock, any other voting securities or any securities convertible or exchangeable into, or any rights, warrants or options to acquire, any such shares, voting securities or convertible or exchangeable securities;

(c) amend its Articles of Incorporation, By-laws or other comparable charter or organizational documents;

(d) develop, acquire or agree to develop or acquire any projects, assets or lines of business, including without limitation by merging or consolidating with, or by purchasing all or a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, joint venture, association or other business organization or division thereof, or any assets that are material, individually or in the aggregate, to ARACOR other than in the ordinary course of business; provided, however, that nothing herein shall prevent the development or acquisition of products in the ordinary course of business;

(e) sell, lease, license, swap, barter, transfer or otherwise convey, mortgage or otherwise encumber or subject to any Lien or prevent from becoming subject to a Lien any of its properties or assets that are material, individually or in the aggregate, to ARACOR;

(f) (1) other than in the ordinary course of business, incur any indebtedness, forgive any debt obligations of any Person to ARACOR, issue or sell any debt securities or warrants or other rights to acquire any debt securities of ARACOR, guarantee any debt securities of another Person, enter into any "keep well" or other agreement to maintain any financial statement condition of another Person or enter into any arrangement having the economic effect of any of the foregoing; or (2) other than advances to customers in the ordinary course of business, make any loans, advances or capital contributions to, or investments in, any other Person, other than travel advances to employees in the ordinary course of business;

(g) fail to file, on a timely basis, including allowable extensions, with appropriate tax authorities all material tax returns required to be filed by or with respect to ARACOR for taxable years or periods ending on or before the Closing Date or fail to remit (or cause to be remitted) any taxes due in respect of such tax returns, except for taxes which are being contested by ARACOR in good faith and for which adequate provision has been made in the consolidated balance sheet and notes thereto on the Balance Sheet Date;

-28-

(h) (1) incur any material obligation to make any payment of, or in respect of, any tax, except in the ordinary course of business; (2) settle any audit by any tax authority, make or change any tax election or file any amended tax returns; (3) agree to extend or waive any statute of limitations on the assessment or collection of tax; or (4) change or adopt any accounting method relating to taxes;

(i) pay, discharge, settle or satisfy any material claims, liabilities or obligations (absolute, accrued, asserted or unasserted, contingent or otherwise), other than the payment, discharge, settlement or satisfaction in the ordinary course of business or reflected or reserved against in the consolidated balance sheet on the Balance Sheet Date or incurred in the ordinary course of business, or, except in the ordinary course of business, waive the benefits of, or agree to modify in any manner, any material confidentiality, standstill or similar agreement to which ARACOR is a party;

(j) except to the extent required under existing plans or agreements or by law as contemplated by this Agreement, increase the compensation payable or to become payable to any of its officers, directors, employees, agents or consultants (other than general increases in wages to employees who are not officers, directors or Affiliates in the ordinary course of business) or enter into or amend any Plan or make any loans to any of its officers, directors, employees, Affiliates, agents or consultants or make any change in its existing borrowing or lending arrangements for or on behalf of any of such Persons pursuant to a employee benefit plan or otherwise;

(k) pay or make any accrual or arrangement for payment of any pension, retirement allowance or other employee benefit pursuant to any employee benefit plan, to any officer, director, employee, consultant or Affiliate or pay or agree to pay or make any accrual or arrangement for payment to any officers, directors, employees, consultants or Affiliates of ARACOR of any amount relating to unused vacation days, except payments and accruals required by law or made in the ordinary course of business; adopt or pay, grant, issue, accelerate or accrue salary or other payments or benefits pursuant to any employee benefit plan with or for the benefit of any director, officer, employee, agent or consultant, whether past or present, other than as required under applicable law or the current terms of any employee benefit plan or in the ordinary course of business; or, except as required by law, enter into any collective bargaining agreement;

(1) make any payments (other than regular compensation payable to officers and employees of ARACOR in the ordinary course of business) or other distributions to, or enter into any transaction, agreement or arrangement with, any of ARACOR's Affiliates, officers, directors, shareholders or their Affiliates, associates or family members or do or enter into any of the foregoing with respect to employees, agents or consultants other than in the ordinary course of business;

(m) except as otherwise permitted by this Agreement modify or amend in a manner adverse to ARACOR the material economic terms of any Material Contract or terminate any Material Contract or waive, release or assign any material rights or claims;

(n) take any action that is intended or may reasonably be expected to result in any of its representations and warranties set forth in this Agreement that are qualified as to materiality being or becoming untrue and those not so qualified being or becoming untrue in any material respect at any time prior to the Closing, or in any of the conditions set forth in <u>Section 10</u> not being satisfied or in a violation of any provision of this Agreement, except, in every case, as may be required by applicable law; or

(o) authorize any of, or commit or agree in writing or otherwise, to take any of the foregoing actions, except as otherwise permitted by

-29-

this Agreement.

9.3.3. Advice of Changes. Each of ARACOR, OSI Subsidiary and OSI shall promptly advise the other party in writing of:

(a) any representation or warranty made by it contained in this Agreement becoming untrue or inaccurate in any material respect;

(b) The failure by it to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied by it under this Agreement;

(c) Any change or event having, or which, insofar as can reasonably be foreseen, would have, a material adverse effect on ARACOR taken as a whole, OSI Subsidiary taken as a whole or OSI taken as a whole or would likely delay or impede the ability of such party to consummate the transactions contemplated by this Agreement or to fulfill its obligations set forth herein and therein;

(d) Any written notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the Merger;

(e) Any written notice or other communication from any governmental authority in connection with the Merger; or

(f) Any action, suits, claims, investigations or proceedings commenced or threatened in writing against, relating to or involving or otherwise affecting ARACOR or OSI that relate to the consummation of the Merger;

provided, however, that no such notification shall affect the representations, warranties, covenants or agreements of ARACOR, OSI Subsidiary or OSI or the conditions to the obligations of ARACOR, OSI Subsidiary or OSI under this Agreement. Additionally, ARACOR shall furnish or make available to OSI, as promptly as practicable following OSI's request therefor, all information reasonably necessary for OSI accurately to determine (i) what amounts, if any, payable under any of the employee benefit plans or any other contract, agreement, or arrangement with respect to which ARACOR may have any liability for federal income tax purposes by virtue of Section 162(a)(1), Section 162(m) or Section 280G of the Code, or (ii) whether ARACOR has entered into any contract, agreement or arrangement that would result in the disallowance of any tax deductions pursuant to Section 280G of the Code.

9.4. No Solicitation. ARACOR shall not, nor shall it authorize or permit any agent, officer, director or employee of, or any investment banker, attorney or other advisor or representative of, ARACOR to directly or indirectly, solicit, initiate, or encourage any inquiries or proposals from, discuss or negotiate with, or provide any non-public information to, any Person (other than OSI or OSI Subsidiary) relating to any transaction involving the sale of any of the assets of ARACOR (other than in the ordinary course of business), or any of the capital stock of ARACOR, or any merger, consolidation, business combination, or similar transaction involving ARACOR.

9.5. **Board Recommendations**. The ARACOR Board shall (i) recommend that the holders of ARACOR Common Stock approve this Agreement and the transactions contemplated hereby to be performed by ARACOR and (ii) cause ARACOR to make commercially reasonable efforts to solicit the approval of the ARACOR Shareholders of this Agreement and the transactions contemplated hereby to be performed by ARACOR. Neither the ARACOR Board nor any committee thereof shall withdraw or modify, or propose to withdraw or modify, in a manner adverse to OSI or OSI Subsidiary, the approval or recommendation by the ARACOR Board or any committee thereof of this Agreement and the transactions contemplated hereby to be performed by ARACOR.

-30-

9.6. Employee Matters.

OSI's benefit plans and policies, including vacation, floating holidays, retirement, severance and welfare plans, shall recognize (i) for purposes of satisfying any deductibles during the coverage period that includes the Closing Date, any payment made by any person employed by ARACOR on or prior to the Closing Date (an "<u>ARACOR Employee</u>") towards deductibles in any health or other insurance plan of ARACOR, and (ii) for purposes of determining eligibility to participate, vesting and for any schedule of benefits based on service, all service with ARACOR, including service with predecessor employers that was recognized by ARACOR and any prior unabridged service with ARACOR ("<u>ARACOR Service</u>"). ARACOR Employees shall transfer, as of the Closing Date, all accrued vacation time earned and determined in accordance with ARACOR's vacation schedule, and shall thereafter accrue vacation time at a rate equal to the rate at which similarly-situated employees of OSI's other Affiliates accrue vacation time. OSI shall recognize each ARACOR Employee's ARACOR Service for purposes of determining eligibility for vacation time following the Closing Date. In addition, OSI shall cause any and all pre-existing condition limitations, eligibility waiting periods and evidence of insurability requirements under any group health plans of OSI in which such employees and their eligible dependents shall participate to be waived and shall provide credit for any co-payments and deductibles prior to the Closing Date. OSI shall recognize ARACOR Employees' accumulated Sick Leave time. Sick Leave means paid time off caused taken in accordance with ARACOR's sick leave policy in effect immediately preceding the Effective Time <u>provided</u>, however, that no further Sick Leave shall accumulate after the Closing.

9.6.1. The Surviving Corporation shall not, and OSI shall cause the Surviving Corporation not to, terminate any ARACOR Employee without cause (as defined in the Key Person Employment Agreements) in the 90 days after the Closing Date.

9.6.2. Other provisions of this <u>Section 9.6</u> notwithstanding, ARACOR Employees who continue to be employed by the Surviving Corporation post-Merger shall accrue employee benefits from and after the Closing only to the extent, and at the rate, that OSI and its other similarly-situated subsidiaries provides such benefits.

10. **OSI SUBSIDIARY'S AND CARDINAL'S CONDITIONS TO CONSUMMATE THE MERGER**. OSI Subsidiary's and OSI's obligations to consummate the Merger on the Closing Date are subject to the satisfaction, at the times set forth below, of each of the conditions set forth in this <u>Section 10</u>. OSI Subsidiary and OSI shall consummate the Merger as soon as practicable following the satisfaction or waiver of the conditions set forth in <u>Sections 10 and 11</u>. OSI Subsidiary or OSI may waive any or all of the following conditions in writing, and any such waiver shall constitute a waiver of the right to receive indemnification under <u>Section 13</u> or any other remedy with respect to such waived condition.

10.1. <u>Accuracy of Representations and Warranties</u>. The representations and warranties of ARACOR and Individual contained herein shall be accurate in all material respects as if made on and as of the Closing Date, as well as on the date when made.

10.2. <u>Performance of Covenants and Deliveries</u>. Individual and ARACOR shall have performed and complied in all material respects with each and all of their obligations and covenants specified in this Agreement that are required to be performed or complied with on or before the Closing Date.

10.3. <u>No Pending Action</u>. No judicial or administrative proceeding shall have been pending on the Closing Date seeking to enjoin or prevent, nor shall any order or injunction have been issued prohibiting, the consummation of the transactions contemplated hereby.

-31-

10.4. <u>Condition of Business and Assets</u>. There shall not have been any material adverse change to the Business, the Assets or ARACOR's financial condition, taken as a whole, since the date of this Agreement.

10.5. <u>Surrender of Certificates</u>. Certificates representing no less than 80 percent of ARACOR Common Stock shall have been surrendered in accordance with <u>Section 2.3</u>.

10.6. <u>ARACOR Stock Options</u>. All ARACOR Stock Options shall have been terminated, and no ARACOR Stock Options shall be outstanding as of the Effective Time.

10.7. <u>Good Standing</u>. ARACOR remains a corporation in good standing, as evidenced by a certificate issued by the Secretary of State of the State of California.

11. **ARACOR'S AND INDIVIDUAL'S CONDITIONS TO CONSUMMATE THE MERGER**. Individual's and ARACOR's obligations to consummate the Merger on the Closing Date, are subject to the satisfaction, at the times set forth below, of each of the conditions set forth in this <u>Section 11</u>. ARACOR shall consummate the Merger as soon as practicable following the satisfaction or waiver of the conditions set forth in <u>Sections 10 and 11</u>. Individual and ARACOR may waive any or all of the following conditions in writing, and any such waiver shall constitute a waiver by Individual or ARACOR of their rights to any remedy with respect to such waived condition.

11.1. <u>Accuracy of Representations and Warranties</u>. The representations and warranties of OSI Subsidiary and OSI contained herein shall be accurate in all material respects as if made on and as of the Closing Date, as well as on the date when made.

11.2. <u>Performance of Covenants and Deliveries</u>. OSI Subsidiary and OSI shall have performed and complied in all material respects with each and all of the obligations and covenants specified in this Agreement that they are required to perform or comply with on or before the Closing Date.

11.3. <u>Cash Payment at Closing</u>. OSI Subsidiary shall have prepared checks for the payment of the Initial Payment for immediate mailing to the recipients of Other Fees, ARACOR Shareholders and ARACOR Optionholders or evidence of wire instructions for wiring of the Initial Payment to the recipients of Other Fees, ARACOR Shareholders, as the case may be.

11.4. <u>No Pending Action</u>. No judicial or administrative proceeding shall be threatened or pending on the Closing Date seeking to enjoin or prevent or claim damages, nor shall any order or injunction have been issued prohibiting the consummation of the transactions contemplated hereby.

12. TERMINATION OF OBLIGATIONS.

12.1. <u>Termination Generally</u>. If the Closing fails to occur on or before the Termination Date, for any reason other than the default or failure of any party hereto to perform any of such party's obligations under this Agreement, then any party to this Agreement may, in such party's sole and absolute discretion, terminate this Agreement by written notice to the other party or parties. Upon such cancellation, each party to this Agreement shall return to the other party or parties all schedules, exhibits, certificates, instruments and documents that had been delivered to such party pursuant to this Agreement.

12.2. <u>Failure to Close By Reason of Default</u>. If the Merger is not consummated by the Termination Date by reason of the default of any party to this Agreement, including, without limitation, the failure of any party to this Agreement to make any delivery required by or to perform any obligation set forth under this Agreement, then upon the occurrence of such breach or default ARACOR, if OSI or OSI Subsidiary is the breaching or defaulting party, or OSI, if ARACOR or Individual is the breaching or defaulting party, may in such non-defaulting party's sole and absolute discretion elect one or more of the following remedies:

-32-

12.2.1. Cancel such party's obligation to perform under this Agreement by providing written notice to the defaulting party or parties of such non-defaulting party's intention to cancel this Agreement, whereby cancellation shall be effective immediately; or

12.2.2. Waive such obligations or conditions, thereby commencing to consummate the Merger.

12.3. <u>Effect of Certain Terminations</u>. Upon any termination that does not result from the breach or default of any party to this Agreement, this Agreement shall be canceled and each party to this Agreement shall thereupon be released from any and all obligations, claims or liabilities in any way arising from or relating to this Agreement; <u>provided</u>, <u>however</u>, that the confidentiality provisions of the Proprietary Information Nondisclosure Agreement dated August 25, 2003 between OSI and ARACOR shall survive.

12.4. <u>Remedies Cumulative</u>. Upon a default prior to the Closing that results in a failure to effect the Closing, a non-defaulting party shall not be entitled to bring any action or exercise any remedy at law or equity arising from the failure to effect the Closing except (a) termination of this Agreement; and (b) an action for damages with respect to the failure by any party to act in good faith in effecting the Closing.

13. INDEMNIFICATION.

13.1. Individual's Indemnifications.

13.1.1. **Breach by ARACOR**. Subject to the limitations set forth in Section 13.2.4, Individual shall indemnify, defend and hold harmless the Surviving Corporation, OSI Subsidiary and OSI, and each of their shareholders, agents, heirs, successors, assigns, partners and joint venturers, from and against any and all Claims and Liabilities that are caused by or relate to the breach by ARACOR of any of its representations, warranties, and covenants that are to be performed by the terms of this Agreement prior to the Effective Time set forth in the ARACOR Documents.

13.1.2. **Breach by Individual**. Subject to the limitations set forth in <u>Section 13.2.4</u>, Individual shall indemnify, defend and hold harmless the Surviving Corporation, OSI Subsidiary and OSI, and each of their agents, heirs, successors, assigns, partners and joint venturers, from and against any and all Claims and Liabilities that are caused by or relate to the breach by Individual of any of his representations, warranties and covenants that are to be performed by the terms of this Agreement prior to the Effective Time set forth in the ARACOR Documents to which Individual is a party.

13.1.3. <u>Claims and Liabilities in Connection with Dissenting Shares</u>. Individual shall indemnify, defend and hold harmless the Surviving Corporation, OSI Subsidiary and OSI, and each of their shareholders, agents, heirs, successors, assigns, partners and joint venturers, from and against any and all Claims and Liabilities that are caused by or relate to any appraisal under Section 1300 et seq. of the CGCL with respect to ARACOR Common Stock to the extent that there is a final, non-appealable determination that any ARACOR Shareholders who have exercised rights under Section 1300 et seq. of the CGCL are entitled to a per share amount in excess of the Merger Payment.

13.2. Certain Matters Relating to Defense and Indemnity.

13.2.1. Definition of "Indemnitor" and "Indemnitee" Notice. Any person seeking defense or indemnity under this Section 13 (hereinafter an "Indemnitee") shall give written notice (each, a "Claim Notice") to any Person (hereinafter the "Indemnitor") from whom such defense or indemnity

-33-

is sought of any Claims and Liabilities that might give rise to a claim by such Indemnitee against the applicable Indemnitor based on the provisions of this <u>Section 13</u>, which notice shall state the nature and basis of such claim and the amount thereof, to the extent the Indemnitee knows such matters. The failure of any Indemnitee to give notice as provided herein shall not relieve the Indemnitor of its obligations under this <u>Section 13</u> except to the extent that the Indemnitor is materially prejudiced thereby.

13.2.2. Duty to Mitigate Damages. Any Indemnitee shall act in good faith and in a commercially reasonable manner to mitigate any Damages they may suffer.

13.2.3. Defense of Claims.

(a) If any Indemnitee receives notice of any claim or legal proceeding by a Person who is not a party to this Agreement (a "<u>Third</u> <u>Party Claim</u>") which is reasonably likely to give rise to any Claim and Liabilities, such Indemnitee shall promptly deliver written notice to the Indemnitor. In connection with any Third Party Claim, the Indemnitor may, upon written notice to the Indemnitee, assume the defense of any such claim or legal proceeding, the costs and expenses of which defense shall be paid by the Indemnitor, if the Indemnitor acknowledges to the Indemnitee in writing the Indemnitor's obligation to indemnify the Indemnitee with respect to all elements of such claim (subject to any limitations on such liability contained in this Agreement). Each Indemnitee shall furnish such information regarding itself or the claim in question as the Indemnitor may reasonably request in writing and as shall be reasonably required in connection with the defense of such claim and litigation resulting therefrom.

(b) If, however, the Indemnitor fails or refuses to undertake the defense of such Third Party Claim prior to the earlier of (i) 30 days after written notice of such claim has been delivered to the Indemnitor by the Indemnitee or (ii) five days prior to the last date of any statutorily-provided time period within which any answer, or other form of response, is be filed with any court of law, the Indemnitee shall have the right to undertake the defense, compromise and settlement of such Third Party Claim with counsel of its own choosing. The Indemnitor shall be entitled to participate in (but not control) the defense of any such action, with its counsel at its own expense and the Indemnitor also shall be entitled to all information reasonably related to a Third Party Claim in Indemnitee's possession or control.

(c) If the Indemnitor assumes the defense of any such claim or legal proceeding, it may use counsel of its choice to prosecute such defense, subject to the approval of such counsel by the Indemnitee, which approval shall not be unreasonably withheld or delayed. The Indemnitee shall be entitled to participate in (but not control) the defense of any such action, with its counsel at its own expense; <u>provided</u>, <u>however</u>, that Indemnitees shall collectively have the right to retain one separate counsel, with the fees and expenses to be paid by the Indemnitor, if representation of such Indemnitee by the counsel retained by the Indemnitor would be inappropriate due to actual or potential different interests between such Indemnitor shall take all reasonable steps necessary to pursue the resolution thereof in a prompt and diligent manner. The Indemnitor shall be entitled to consent to a settlement of, or the stipulation of any judgment arising from, any such claim or legal proceeding, with the consent of the Indemnitee, which consent shall not be unreasonably withheld or delayed.

(d) Notwithstanding the foregoing, however, the Indemnitee shall be entitled to the control of the defense or any such action if it is reasonably likely to result in liabilities which, taken with other then existing claims by any other Indemnitee under this <u>Section 13</u>, would not be fully indemnified hereunder, unless the amount that would be indemnified hereunder is more than 50% of the likely amount of liabilities in such action. If the Indemnitee is entitled to control the defense of an action, the Indemnitee shall be entitled to consent to a settlement of, or the stipulation of any judgment arising from, any such claim or legal proceeding, with the consent of Indemnitor, which consent shall not be unreasonably withheld or delayed. The party controlling the defense of a Third Party Claim shall keep the other parties fully informed of the defense of any such Third Party Claim.

-34-

13.2.4. Limitations Upon Indemnifications.

(a) Subject to the provisions of the Individual Indemnification Agreement, the obligations of Individual under this <u>Section 13</u> to pay Indemnitees in indemnification against Claims and Liabilities shall not exceed 20% of the aggregate amount of Merger Consideration made pursuant to the terms of this Agreement, (the "<u>Indemnity Cap</u>").

(b) Neither OSI nor the Surviving Corporation shall deliver a Claim Notice under this Section 13 unless and until the aggregate amount of Claims and Liabilities with respect to which Individual is obligated under this Section 13 to indemnify OSI and OSI Subsidiary, combined, equal or exceed \$250,000 (the "Basket Amount"). Once the Basket Amount has been satisfied and Indemnitee is permitted to submit a Claim Notice, the Indemnitee shall be entitled to claim indemnification for the full amount of such claim (including the Basket Amount) according to the terms of this Section 13.

13.2.5. <u>Remedies Exclusive</u>. Except in instances of willful misrepresentation or intentional fraud (including any such misrepresentation or fraud by ARACOR), recourse pursuant to the provisions of this <u>Section 13</u> shall be the sole and exclusive remedy for claims against ARACOR and Individual of all parties to this Agreement for Claims and Liabilities.

13.2.6. Survivability. The indemnities and defense obligations and other covenants and agreements set forth in this Section 13 shall survive any termination of this Agreement pursuant to Section 12 above, and shall survive the Closing and the consummation of the Merger; provided, however, that the defense and indemnity obligations under this Section 13 relating to Claims and Liabilities caused by or relating to breaches of representations and warranties set forth in Sections 6 through 7 of this Agreement shall terminate on the first anniversary of the Closing Date, except with respect to any Claims and Liabilities covered by a Claim Notice that the Indemnite has delivered to the Indemnitor prior to the first anniversary of the Closing Date. Notwithstanding the preceding sentence, the defense and indemnity obligations under this Section 13 relating to Claims and Liabilities caused by or relating to breaches of representations and warranties set forth in Sections 6.12, 6.17, 6.25 and 7.2 of this Agreement shall continue until the third anniversary of the Closing Date, and Claims and Liabilities caused by or relating to breaches of representations and warranties set forth in Sections 6.12, 6.17, 6.25 and 7.2 of this Agreement shall continue until the third anniversary of the Closing Date, and Claims and Liabilities caused by or relating to breaches of representations and warranties set forth in Sections 6.12, 6.17, 6.25 and 7.2 of this Agreement shall continue until the third anniversary of the Closing Date, and Claims and Liabilities caused by or relating to breaches of representations and warranties set forth in Section 6.8 shall continue until the expiration of the statutes of limitation applicable to the subject matter from which such Claims and Liabilities arise, at which time such obligations shall terminate, except with respect to any such Claims and Liabilities covered by a Claim Notice that the Indemnitee has delivered to the Indemnitor prior to the third anniversary of the Closing Date or the expiration

14. SHAREHOLDER REPRESENTATIVE.

14.1. <u>Appointment of Shareholder Representative</u>. By virtue of the adoption and approval of this Agreement by the requisite shareholder vote, each of the shareholders shall be deemed to have agreed to appoint a shareholder representative (the "<u>Shareholder Representative</u>") to act as their agent and attorney-in-fact, for and on behalf of the ARACOR Shareholders to give and receive notices and communications, to agree, to negotiate, to enter into settlements and compromises of, and demand arbitration and comply with orders of courts and awards of arbitrators with respect to such claims, in each case relating to this Agreement or the transactions contemplated hereby (including, without limitation, claims arising from the Contingent Consideration), and to take all other actions that are either (i) necessary or appropriate in the judgment of Individual for the accomplishment of the foregoing or (ii) specifically mandated by the terms of this Agreement. The Shareholder Representative shall be Individual for the purposes of this <u>Section 14.1</u> and elsewhere in this Agreement as applicable. Such agency may be changed by the ARACOR Shareholders

-35-

from time to time upon not less than thirty (30) days prior written notice to OSI and OSI Subsidiary; provided, however, that the Shareholder Representative may not be removed unless holders of three quarters of the shares of ARACOR Common Stock agree to such removal and to the identity of the substituted agent.

14.2. Limitation of Liability; Reimbursement of Expenses. No bond shall be required of Individual, and no compensation shall be received for services rendered by Individual, in his capacity as Shareholder Representative. The Shareholder Representative shall not be liable for any act done or omitted hereunder as shareholder representative while acting in good faith and in the exercise of reasonable judgment and any act done or omitted pursuant to the advice of counsel shall be conclusive evidence of such good faith. ARACOR Shareholders shall severally indemnify and hold Individual harmless against any loss, liability or expense incurred without gross negligence or bad faith in his capacity as Shareholder Representative and arising out of or in connection with the acceptance or administration of his duties as Shareholder Representative. OSI and OSI Subsidiary acknowledge that Individual may have a conflict of interest with respect to his duties as shareholder representative, and in such regard Individual has informed OSI and OSI Subsidiary that he will act in the best interests of ARACOR Shareholders. To the extent that disputes arise related to the payment of the Merger Consideration, the Shareholder Representative shall be entitled to reimbursement of fees and costs of legal counsel pursuant to <u>Section 4.6.3</u>.

14.3. <u>Access to Information</u>. Shareholder Representative, in accord with <u>Section 14.1</u> hereof, shall have reasonable access to information about ARACOR and the Business and shall receive the reasonable assistance of OSI's and OSI Subsidiary's officers and employees for purposes of performing his duties and exercising his rights hereunder, provided that the Shareholder Representative shall treat confidentially and not disclose any nonpublic information from or about OSI, OSI Subsidiary, ARACOR or the Business to anyone except on a need to know basis to individuals who agree to treat such information confidentially.

15. GENERAL PROVISIONS

15.1. <u>Amendment, Modification and Waiver</u>. This Agreement may not be amended, modified or supplemented except pursuant to an instrument in writing signed by each of the parties hereto, except that any party to this Agreement may waive any obligation owed to such party by another party under this Agreement, provided such waiver is in writing. The waiver by any party hereto of a breach of any provisions of this Agreement shall not operate or be construed as a waiver of any subsequent breach.

15.2. <u>Severability</u>. If any provision of this Agreement as applied to any party or to any circumstance shall be found by a court of competent jurisdiction to be void, invalid or unenforceable, the same shall in no way affect any other provision of this Agreement, the application of any such provision in any other circumstance, or the validity or enforceability of this Agreement, and any provision that is found to be void, invalid or unenforceable shall be curtailed and limited only to the extent necessary to bring such provision within the requirements of the law.

15.3. Entire Agreement. This Agreement and the confidentiality provisions of the Proprietary Information Nondisclosure Agreement between ARACOR and OSI dated August 25, 2003 contain the entire agreement among the parties hereto with respect to the transactions contemplated hereby and supersede all prior agreements or understandings among the parties with respect thereto.

15.4. <u>Attorneys' Fees</u>. In the event of any litigation or arbitration between or among the parties hereto respecting or arising out of this Agreement, the successful or prevailing party shall be entitled to recover his reasonable attorneys' fees and other costs in connection therewith, including, without limitation, any attorneys' fees incurred after a judgment has been rendered by a court of competent jurisdiction. For the purposes hereof, the prevailing party shall be deemed to be the prevailing party if such party is awarded more than half of the amount in dispute plus any amounts not in dispute.

-36-

15.5. <u>Further Acts</u>. Each party, upon the reasonable request of any other party, shall execute and deliver all such further instruments, documents and papers, and shall perform any and all acts necessary to give full force and effect to all of the terms and provisions of this Agreement (so long as such performance does not involve more than \$5,000 expense in the aggregate to OSI and OSI Subsidiary on the one hand, or to ARACOR and Individual on the other hand, or the requesting party pays the expenses incurred).

15.6. <u>Descriptive Headings</u>. The paragraph and section headings in this Agreement are for convenience only and shall not control or affect the meaning or construction of any provision of this Agreement.

15.7. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, and each of which shall be deemed an original, but all such counterparts together shall constitute but one agreement.

15.8. <u>Binding Effect</u>. All the terms and provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. Anything contained herein to the contrary notwithstanding, this Agreement shall not be assignable by any party hereto without the consent of the other parties hereto.

15.9. <u>Third Party Beneficiaries</u>. No person shall be a third party beneficiary of this Agreement and no person other than the parties hereto and their permitted successors and assigns shall receive any of the benefits of this Agreement; <u>provided</u>, <u>however</u>, that ARACOR Shareholders and ARACOR Optionholders shall have the right to pursue legal action against OSI and/or OSI Subsidiary in the event of a breach of this Agreement by OSI or OSI Subsidiary following the Closing, through the use of the Shareholder Representative.

15.10. **Notices**. All notices, statements and other documents that any party is required or desires to give to any other party hereunder shall be given in writing and shall be served in person by express mail, by certified mail, by overnight delivery, or by facsimile at the respective addresses of the parties as set below, or at such other addresses as may be designated in writing by such party in accordance with the terms of this paragraph.

If to ARACOR:

352 Java Drive Sunnyvale, CA 94089 Attention: President

With a copy to:

Gray Cary Ware & Freidenrich LLP 2000 University Avenue East Palo Alto, CA 94303-2248 Attention: Diane Holt Frankle, Esq.

-37-

If to Individual, at the address set forth on the signature pages hereto.

If to OSI Subsidiary or OSI:

12525 Chadron Avenue Hawthorne, CA 90250 Attention: President

With a copy to:

Richman, Mann, Chizever, Phillips & Duboff 9601 Wilshire Boulevard, Penthouse Beverly Hills, CA 90210 Attention: Gerald Chizever

Delivery shall be deemed conclusively made (i) at the time of service, if personally served, (ii) 3 days following the date deposited in the United States mail, properly addressed and postage prepaid, if delivered by express mail or certified mail, (iii) 1 day following the deposit with the private overnight deliverer, if served by overnight delivery, and (iv) at the time of electronic transmission (as confirmed in writing), provided a copy is mailed within twenty-four (24) hours after such transmission. The time to respond to any notice shall run from the time the notice is deemed hereunder to have been delivered to the person to whom the notice is addressed.

15.11. <u>Applicable Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of California (without regard to conflicts of laws principles), and the parties hereby consent to the jurisdiction of California state courts or federal courts located within California over all matters relating to this Agreement.

15.12. **Plural Includes Singular**. Except as indicated in the next sentence, whenever used in this Agreement, the singular shall include the plural and the plural shall include the singular. Without limiting the generality of the foregoing, when used in this Agreement the term "Individual" refers to each of the Persons constituting Individual; <u>provided</u>, <u>however</u>, that the foregoing shall not impose joint and several obligations on Individual except as expressly set forth in this Agreement.

15.13. <u>Expenses</u>. Except as otherwise expressly provided herein, each party to this Agreement shall pay its own costs and expenses in connection with the transactions contemplated hereby.

15.14. **Gender**. References to the masculine gender in this Agreement shall include the feminine and neuter genders; references to the singular shall include the plural; and references to "person" shall include any Person; all as required by the context of this Agreement.

15.15. <u>Announcements</u>. Prior to the Closing, no party to this Agreement shall issue or cause the publication of any press release or other announcement with respect to this Agreement or the Merger without the consent of the other party, except where such release or announcement is required by applicable law.

15.16. Incorporation By Reference. All Schedules and Exhibits hereto are incorporated herein by this reference.

[signatures on following pages]

-38-

IN WITNESS WHEREOF, each of the parties has caused this Agreement to be duly executed all as of the date first written above.

ADVANCED RESEARCH & APPLICATIONS CORPORATION

By:

Its:

ROBERT A. ARMISTEAD

_

Individual Address:

OSI SYSTEMS, INC.

- By: Ajay Mehra
- Its: Executive Vice President

OSI SUBSIDIARY, INC.

By: Ajay Mehra Its: Executive Vice President